



Review Survey on e-Powered Micro Personal Mobility Vehicles: Exploring Current Issues towards Future Developments

Stefania Boglietti, Benedetto Barabino * D and Giulio Maternini

Department of Civil, Environment, Land and Architecture Engineering and Mathematics (DICATAM), University of Brescia, Via Branze 43, 25123 Brescia, Italy; s.boglietti001@unibs.it (S.B.); giulio.maternini@unibs.it (G.M.) * Correspondence: benedetto.barabino@unibs.it

Abstract: Nowadays, the diffusion of electric-powered micro Personal Mobility Vehicles (e-PMVs) worldwide—i.e., e-bikes, e-scooters, and self-balancing vehicles—has disrupted the urban transport sector. Furthermore, this topic has captured many scholars and practitioners' interest due to multiple issues related to their use. Over the past five years, there has been strong growth in the publication of e-PMV studies. This paper reviews the existing literature by identifying several issues on the impact that e-PMVs produce from different perspectives. More precisely, by using the PRIMA's methodological approach and well-known scientific repositories (i.e., Scopus, Web of Science, and Google Scholar), 90 studies between 2014 and 2020 were retrieved and analyzed. An overview and classification into endogenous issues (e.g., impact on transport and urban planning) and exogenous issues (e.g., impact on safety and the environment) are provided. While several issues are deeply investigated, the findings suggest that some others need many improvements. Therefore, the status quo of these studies is being assessed to support possible future developments.

Keywords: micromobility; electric scooter; e-scooter; personal mobility vehicle; personal transporter; segway; micromobility problems

1. Introduction

Nowadays, the increasing use of private transport has had multiple effects on the urban space (for parking lots, streets) and its livability, thus presenting the account of its unsustainability [1]. Moving and parked cars occupy valuable urban spaces useful for citizens. Therefore, traffic on the roads should be adapted to the existing urban road space with the help of public transport. Indeed, several guidelines have been developed and promoted by the European Union to encourage public transport [2,3]. The unsustainability of urban transport is also affected by urban freight transport. Although urban freight transport is essential to meet citizens' needs, it has led to greater concern for the global and local environment (e.g., air pollution, noise, and vibrations) and safety and security issues [4,5]. Thus, poor air quality, traffic congestion, and the growth of road crashes push towards the rising need for alternative urban mobility solutions. Hence, academics, mobility experts, and urban planners are trying to rethink people's transport mode selections by investigating less energy-intensive modes such as walking and the use of micromobility devices [6]. Although much has been written about pedestrian mobility (definitions, methods, etc.), micromobility is a recent topic, and a common accepted definition misses from the literature. In the United States, micromobility refers to vehicles with a mass of no more than 350 kg and a design speed of no more than 45 km/h. In Europe, there is no univocal definition of the term, but the European Union regulation No. 168/2013 has established the L-category vehicles as a reference for the member countries. L-category vehicles are two-, three- and four-wheeled motor vehicles. The category uses power, power source, speed, length, width, and height as classification criteria [7]. Generally speaking, micromobility refers to electric-powered micro Personal Mobility Vehicles (e-PMVs). These



Citation: Boglietti, S.; Barabino, B.; Maternini, G. Survey on e-Powered Micro Personal Mobility Vehicles: Exploring Current Issues towards Future Developments. *Sustainability* **2021**, *13*, 3692. https://doi.org/ 10.3390/su13073692

Academic Editor: Antonio Comi

Received: 5 March 2021 Accepted: 24 March 2021 Published: 26 March 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). devices include general electric scooters (or e-scooters), e-bikes, and self-balancing vehicles (in this paper, e-PMVs include only electric scooters as the e-kick scooter, Segway, hoverboard, and monowheel. Therefore, micro vehicles that can be driven standing). In the e-scooter category, the e-kick scooter is one of the most popular due to its ease of use and handling. Segway is a two-wheeled electric device that exploits the presence of the handlebar to facilitate correct posture, balance, and safe driving. Hoverboard and monowheel are one-wheeled self-balancing vehicles and exploit the weight sensors by tilting the body forward to start and backwards to brake; hence, they require the right balance and much practice to be used.

The e-PMVs are gaining popularity as an environmentally friendly transport mode in urban contexts, and their use results in several benefits. Indeed, switching towards e-PMVs may increase community relationship, possible reduction of traffic congestion (for short trips) and emission levels, and improvement of the air quality. Moreover, they can be used privately or docked- or dockless-shared. A docked device has a specific location where it can be picked up and released, while a dockless device has no fixed home location and is dropped and picked up anywhere. Finally, the e-PMVs market is expected to grow at least until the year 2024 at a compound annual growth rate of 7.0% [8].

The diffusion of e-PMVs in the United States since 2017 and in several European metropolises (e.g., Barcelona, Milan, and Paris) since 2018 has raised several issues related to transport and urban planning, safety, and environment that disrupted the urban transport field and captured the attention of many scholars and practitioners. Therefore, several questions should be answered.

On the one hand, since the e-PMVs could provide an alternative transport solution, what is the trip pattern they follow (i.e., where, when, why, how they are adopted)? What quota of trips can they satisfy? Who are the users and the reasons that lead to their use? For instance, users prone to use e-PMVs devices praise their ability to provide a comfortable and fast trip, but above all their ability to make travel joyful: freedom and driving control, combining the pleasure of walking with the excitement of cycling and the comfort of skateboarding [9]. Moreover, what is the user's behavior while driving and/or parking e-PMVs? These devices can be admitted in public spaces like roads, squares, and parks. Therefore, what is the impact of them on e.g., shared infrastructures? To what extent are these new vehicles accepted by other users? How can their circulation be regulated? For instance, before enabling e-PMVs to circulate on shared infrastructures, it would be useful to assess the impact of these new vehicles on public spaces and on other road users.

On the other hand, the e-PMVs are small and light vehicles (foldable and manageable). This implies that they may conflict with other road users while on the move, hindering their transit, causing serious crashes. Therefore, road safety must be considered when introducing new vehicles. Evidence from the USA shows that crashes with e-PMVs cause more injuries than all other devices because users have to stand while driving, they move at relatively high speed (if compared to walking speed) and because no driving license or experience is required for their use [10]. Therefore, what are the effects of the crashes? What are the consequences when pedestrian and e-PMVs conflict? What are the behaviors of safe driving? For instance, it would be important to understand the psychophysical conditions of drivers and their impact when conflicts with other road users occur (especially pedestrians). Finally, e-PMVs are defined as sustainable transport because they have zero emissions. However, if the entire life cycle is considered, can they be still defined as sustainable? For instance, unfavorable users to e-PMVs question sustainability [9].

By scrutinizing existing literature, this paper aims to answer all these questions to provide an overview and classification of current knowledge on e-PMVs and suggests a possible research agenda. Despite the emerging interest in the research of e-PMVs, to the best of our knowledge, there are no detailed surveys investigating all the previous questions. O'Hern & Estgfaeller [11] provided a scientometric review to synthesize, sort rapidly, and analyze bibliographic data and display micromobility knowledge. However, the issues encountered with the introduction of e-PMVs regarding transport, urban planning,

road safety, and environment were not investigated in detail. Moreover, they refer to micromobility as intended in the USA, whereas this paper focuses on e-PMVs according to the European definition. Therefore, this survey was carried out to be useful to public administrations and vehicle providers, in addition to the academia.

The remaining paper is organized as follows. Section 2 shows the methodology used, including the search strategy and type of analysis. Section 3 presents relevant descriptive statistics and introduces the issues investigated. Section 4 briefly analyzes the 90 publications deemed suitable. Section 5 provides insights for some future research developments. Section 6 concludes this survey.

2. Methodology

This paper follows the methodological approach proposed by Cooper [12] and Moher et al. [13] and applied in Barabino et al. [14]. It is organized into five stages:

- 1. Formulation of problem and research questions (see Section 1).
- 2. Definition of a data search strategy, including multiple channels to avoid biases in coverage.
- 3. Keywords evaluation and selection of retrieved data, including the selection criteria of suitable data.
- 4. Analysis and interpretation of the literature, including statistics about sources, number of retrievals and literature finally reviewed (see Section 3).
- 5. Results with a brief comment on each paper (see Section 4).

A computerized search strategy was adopted for the sake of fastness and efficiency. Scopus and Web-of-Science have been used as they are the largest abstracts and citations database of literature. Since these databases cannot contain all references, a separate search on Google Scholar was performed. It is a freely accessible web search engine that indexes the full text or metadata of scholarly literature across various publishing formats and disciplines. Despite providing a wealth of information, a free web search is disregarded due to the abundance of non-scientific information. Other papers have been retrieved using the references cited in the literature already studied (i.e., ancestry approach) and informal contacts.

The search included studies from 2014 to July 2020. The original review was updated due to the publication of recent papers. The same search and selection strategy was applied for the months of August 2020 to December 2020. The publications retrieved for these years are listed in Appendix A, but these papers were not reviewed. The choice to examine articles starting in 2014 was made because research on e-PMVs was not common before.

The papers search is based on the title, abstract and keywords. Several search terms are adopted, like combinations of keywords to cover different idioms of e-PMVs. Studies on "Micromobility", "Electric Scooter", "E-scooter", "Personal Mobility Vehicle", "Personal transporter", and "Segway" were selected. The selection criteria of these keywords are as follows. "Micromobility" represents the category of e-PMVs to cover short distances in the European context. "Electric scooter" and its abbreviation "E-scooter" are the terms with which these vehicles are generally referred. Other related terms are "Personal Mobility Vehicle" or "Personal Transporter". "Segway" is chosen because it defines a specific category of e-PMVs. The search was also carried out for the other types of e-PMVs, i.e., "hoverboards" and "monowheels" (the search for hoverboards only has produced studies referring to pediatric crashes because they are mainly used by children as toys. Therefore, these studies were excluded. In addition, the search for monowheels did not produce results). Unlike Scopus and Web-of-Science, Google Scholar allowed a search based on the title and along with the text. Due to the abundance of papers retrieved searching in the text, the only search criteria used in Google Scholar was the title.

The analysis and interpretation of the literature on e-PMVs may differ by country or area. Therefore, the "country of affiliation" (the workplace of scientists, not their nationality) and "country in which research is applied" are the geographical indicators retrieved from

the analyzed literature. Conversely, "the number of scientists involved per country" and "the number of publications per year" quantified the impact of research.

Finally, the current knowledge base was assessed by analyzing the main issues to provide some possible contents of a research agenda.

3. Results

3.1. General Statistics about the Survey

According to PRISMA's statement [13], Figure 1 shows that 1486 papers were identified by database searching. Preliminary results showed two main categories in which the research on e-PMVs can be divided. The first focused on their mechanics and their components. The second category focused on the impacts e-PMVs have on transport, urban planning, road safety, and the environment. The review considers only studies of the last category because they are closer to the research field of practitioners and academic experts in mobility and urban planning. Consequently, 1127 articles were sooner removed after screening the title and 359 remains. An additional 278 papers were collected from an ancestry approach and informal contacts. After the removal of duplicates, 276 papers were included. Next, 175 papers were excluded after the screening of the abstract, as they were not directly focused on the impact of e-PMVs on urban planning and transportation (67), were not written in a language understood by the authors—i.e., Finnish and Chinese—(13), were not available for consultation (42), and were retrieved from commercial magazines, technical reports, and press releases that miss of the research background (53). Finally, only English publications in journal articles and conference papers were retrieved. Therefore, 101 papers were evaluated in the full text, and 11 were excluded since they discuss different topics from those chosen. Finally, 90 publications were reviewed: 69 in journals, 18 in conferences' papers, 1 meeting abstract, 1 position paper, and 1 report summary. The most frequent journals were Journal of Transport Geography, Sustainability and Transportation Research Part D (3 times). Recurrent journals were also Accident Analysis & Prevention, EMA—Emergency Medicine Australasia, Journal of Cleaner Production, Singapore medical journal, and Transport Findings and Transportation Research Record with two publications, respectively. These different outlets showed the fragmentation of the journals addressing e-PMVs for a varied audience.

We retrieved 47 unique studies from Scopus, 8 from Web of Science, 19 from Google Scholar, 14 by the ancestry approach, and 2 informal contacts.

Figure 2 clearly shows that micromobility is an emerging research area over the last years. From 2016 to 2019, the number of publications quintupled, and it was still growing in 2020.



Figure 1. Flow diagram to identify the papers reviewed according to PRISMA's statement.



Figure 2. Number of publications reviewed per year: period 2014–July 2020.

Table 1 reveals the geographical distribution of the publications reviewed. The first column reports the name of the continent (in bold) and the related countries.

| Country ^a | Number of Researchers Involved ^b | Number of Publications ^c | Country to Which Research Applies ^d |
|----------------------|--|--|---|
| Asia | 67 | 20 | 13 |
| China | 7 | 2 | 1 |
| Japan | 20 | 5 | 3 |
| Korea | 5 | 1 | 1 |
| Saudi Arabia | 1 | 1 | - |
| Singapore | 21 | 5 | 5 |
| Taiwan | 12 | 5 | 3 |
| Vietnam | 1 | 1 | - |
| Europe | 73 | 26 | 28 |
| Austria | 9 | 3 | 5 |
| Belgium | 6 | 1 | 1 |
| Denmark | 4 | 1 | 3 |
| Finland | 3 | 1 | 1 |
| France | 2 | 2 | 4 |
| Germany | 26 | 5 | 6 |
| Greece | 1 | 1 | - |
| Italy | 4 | 1 | 2 |
| Lithuania | 2 | 1 | - |
| Norway | 2 | 2 | 1 |
| Russia | 1 | 1 | - |
| Spain | 3 | 1 | 1 |
| Sweden | 2 | 2 | 1 |
| Switzerland | 3 | 1 | 2 |
| UK | 5 | 3 | - |
| Not specified | - | - | 1 |
| North America | 175 | 45 | 67 |
| Canada | 14 | 7 | 2 |
| USA | 161 | 38 | 65 |
| Oceania | 25 | 8 | 12 |
| Australia | 10 | 3 | 3 |
| New Zealand | 15 | 5 | 9 |
| Not specified | 1 | 1 | 12 |
| Total | 341 | 100 | 132 |

Table 1. Geographical distribution of the publications reviewed.

- No data are reported for the considered column. ^{a.} The authors are aware that the identified literature is not exhaustive of all the documents on e-PMVs, but it is driven by applying the previous search strategy. Additionally, some retrieved technical reports and press releases were excluded because the research background had been lacking. For instance, some reports can be available in other countries, but they were omitted according to our classification criteria. ^b Based on affiliation, not on the nationality of the researchers. Authors who wrote more than one paper are counted once. ^c >90 publications reviewed because some articles are written by authors from different countries. ^d Some studies refer to many countries. If some studies refer to the same country, it was considered as a separate study.

The second column shows that the research was carried out by 51% of researchers in North America, 21% in Europe, 20% in Asia, and 8% in Oceania. In the third column, the number of publications shows that North America, Europe, and Asia Europe provided 45%, 26%, and 20% publications, respectively. Other continents such as Africa, South America, and Oceania are barely or not at all covered by the literature we reviewed.

The last column reports the number of studies considering the country in which the research was conducted. The distribution of studies is as follow: North America 51%, Europe 21%, Asia 10%, and Oceania 9%. Another 10% is related to countries not specified.

3.2. e-PMVs Research Issues

The literature can be classified according to problems, methods, issues, etc. This paper presents a classification based on some research issues without neglecting the interrelationships between them. These issues may be clustered according to two lines of

research, based on the different impacts of e-PMVs over the urban context: (1) endogenous issues concerning the impact on transport and urban planning, and (2) exogenous issues with respect to the indirect impact on road safety and the environment. The endogenous issues refer to problems strictly related to the use of e-PMVs in the public space, while the exogenous issues refer to the external effects of their use, therefore the impact on users' road safety and on the environment.

This classification is motivated by the following reasons.

First, the introduction of e-PMVs on a consolidated transportation system requires revisiting some "traditional" patterns related to transport and urban planning that may not be applicable to the new mode. The literature is studying the main pattens to understand where and when these vehicles are used, what transport system they replace, and users' type who use them the most. In addition, the movement of these vehicles in public spaces creates conflicts with other road users. Therefore, studies are looking at how the city regulates and organizes public space and how drivers behave while parking and using e-PMVs.

Second, e-PMVs may be subject to serious crash risks, owing to the small size and lightness. The consequences of crashes with these vehicles are often far more severe than other road crashes. Therefore, a growing literature is investigating the road safety conditions associated with the use of these devices. It includes studies on the effects of crashes for the user her/himself, the consequences of other vulnerable road users involved (e.g., pedestrians and cyclists), and some facets related to safe driving. Moreover, e-PMVs are believed to be a sustainable alternative to using cars for short trips. Therefore, other literature is investigating the environmental impact associated with its use.

The classification is structured as shown in Figure 3. In particular, the division of endogenous and exogenous problems is shown and, subsequently, the problems outlined. Furthermore, in Figure 3 the reference paragraph and the main topics addressed in this survey have been indicated.

This survey shows that the research is not evenly distributed between North America and Europe. Research objectives change in different countries due to the (differentiated time) development of e-PMVs. North America dominates the area in terms of road safety and urban and transport impact, while Europe dominates environmental impact. It should be noted that the greater attention of Europe to such impacts compared to North America could be explained by the implementation of major environmental policies [15]. Finally, in Asia and Oceania, there is a balanced distribution of studies on road safety and planning.



Figure 3. Structure flow chart.

4. Current Knowledge Base

This section presents the two lines of research as retrieved from the literature. All studies are summarized by tables, including the type of study, sample size, data sources, the period covered, analytical tool, and relevant insights. For instance, the sample size could help understand the impact of the study from a practical perspective, and the relevant insights help to understand the peculiarities of each study.

The types of study are distinguished among qualitative, quantitative, descriptive, and theoretical. Qualitative studies analyze the behavior and motivations behind the use of e-PMVs. Instead, quantitative ones are based on descriptive and statistics models and examine various data on e-PMVs. Descriptive ones discuss general information on

how to address the impact of e-PMVs. Theoretical studies formulate models without experimentation in real case studies.

Data sources are very similar and usually include automatic data and surveys. Automatic data can be not specified, collected via the Application Programming Interface (API), smartphone or smartwatch applications, and weight sensors. Surveys usually concern questionnaires and/or personal interviews with providers and users. Finally, reference is made to on-site observations, with manual data collection, for observations without further specifications.

Analytical tools for qualitative studies include synthesized and encoded textual interviews, travel diaries, and survey. In quantitative studies, descriptive models (simple percentages, cluster analyzes, etc.), inferential models (linear regressions, structural equation models, etc.) and optimization methods are used. Descriptive studies adopt a qualitative description, while theoretical studies present different types of models.

Finally, the main findings of each study are briefly discussed.

4.1. Endogenous Issues: Impact on Transport and Urban Planning

Forty-four articles address the impact of e-PMVs on transport and urban planning. Real data and observations guide them to understand where, when, and how these devices are adopted (trip pattern). They also examine the motivations that lead users to use these new devices, the modal shift generated introducing e-PMVs, and attempting to profile users by several methods. Finally, some studies analyze the user's driving and parking behavior with these devices and the facets of the regulation and design of urban public spaces needed to accommodate e-PMVs.

4.1.1. Trip Pattern

Many studies investigated the trip patterns of the e-PMVs (Table 2). They largely showed the main characteristics of a trip in terms of average length traveled, time, and speed of the journey. Moreover, they reported where and when e-PMVs were used, including the diversity of land use, and the climatic conditions of the city. Finally, the main reasons that encourage their use were also reported.

Key findings concern trip characteristic of PMVs, which seem especially useful for short trips. Indeed, almost of studies showed that the average trips length is from 1.2 km to 2.7 km, the average time is from 10 min to 16 min, and the average speed is from 7 km/h to 10 km/h [16,21,24–26,33,36]. The speed differences might depend on the purposes of the trips. For instance, Almannaa et al. [16] and Hardt & Bogenberger [21] showed that users tend to drive at a lower average speed for recreational purposes, such as shopping or leisure, than for commuting or errands. Moreover, they travel at a slower speed than e-bikes. Hardt & Bogenberg [22] enforced the 2017 results in highlighting the advantages and disadvantages of using e-PMVs. The charging infrastructure and parking are the most obvious advantages, while the weather conditions, luggage restrictions, and road safety are disadvantages. Although it is claimed that they are mainly used for short trips, Markvica, Schwieger & Aleksa [26] argued that it is not clear whether these also cover first/last mile distances. Indeed, they show that e-PMVs could be a good solution, but it is necessary to focus on the user group of individual motorized transport.

Several studies investigated where and when e-PMVs were adopted. These studies agree that e-PMVs are mainly used near the city center (downtown, close to universities and/or university campuses), where there is greater access to multimodal transit (e.g., bus or metro) and greater land-use diversity. In addition, travel to/from bus stops or parking lots is also studied as a last-mile journey. The use is likely to begin and end in residential, commercial, and industrial areas [18,20], and e-kick scooters are the favorite devices [17,18,24]. Furthermore, Hawa et al. [23] pointed out that e-kick scooters are very often available near bike paths, while Jiao & Bai [24] noticed that the further away users are from these places, the less likely they are to travel by e-kick scooter. Contrasting results were also obtained looking at the day of the use of e-PMVs. Caspi, Smart & Noland [18] showed

that they are used more on weekends and holidays, whereas Hawa et al. [23] on weekdays and during the afternoons and the evening [20,27,29,34,36]. Bai & Jiao [17] found these differences by comparing Austin's and Minneapolis's cities in the USA: in Austin, e-PMVs are used more on weekends, while in Minneapolis during the week. Mathew et al. [27] and Zou et al. [36] showed that e-kick scooter trips are mostly concentrated during the central daytime hours, followed by the evening rush hours. Specifically, between 4 pm and 9 pm during the week and between 2 pm and 7 pm on weekends.

The tendency to use e-PMVs most during the afternoon and evening may also depend on the climatic conditions of the city. Mathew, Liu & Bullock [28] showed how meteorological variables (i.e., amount of precipitation, snowfall, wind speed, visibility, and average temperatures) significantly affected the number of trips per hour (30%–80% in winter months). A quite similar result is pointed out by Noland [33] and Hardt & Bogenberger [21]; the latter also added the scarce use of clothing suitable for the climate. Hawa et al. [23] showed an association with the daily temperature: e-PMVs are used more in the afternoon (because the temperature is high) and during humid days, while during rainy days their use decreases. Finally, the temperature also affects the trip length performed by e-PMVs: higher average temperatures increase it, rain and snow reduce it, while stronger winds slightly reduce it [33].

A crucial issue of the trip pattern is to investigate the main reasons for using these devices. Tuncer & Brown [35] reported the feeling of freedom and fun while using them. In addition, the reduction in trip time is another key motivation in situations of need; therefore, they can be an alternative to public transport. The lack of human effort required to drive an electric scooter makes it a preferable and viable transport vehicle over a bicycle or skateboard. Moreover, e-PMVs are foldable and easily transportable and can be stored indoors to protect them from theft or vandalism. These results suggest that they may be adopted for both work and leisure trips [16,20,21,25,30]. Espinoza et al. [20] added that e-PMVs are used for business-to-business trips, business trips to/from parking, and for reaching bars or restaurants. Nevertheless, some studies show that among the various soft transport system available (e-PMVs, docked or dockless bikes and e-bikes), e-PMVs are mainly used for leisure and non-commuter travel. The large mid-day e-PMV travel concentration observed over the weekend supports this fact [30,34,36]. Unlike other studies Davies, Blazejewski, & Sherriff [19] showed how e-PMVs are useful for tourist tours, as they help to make tourist places less congested, more sustainable, and more desirable.

For trips to/from work or commuting, bike-sharing and e-bike sharing systems are preferred [30,34]. According to Reck et al. [34] devices' choice depended on trip-distance: dockless e-kick scooters for very short trips, docked bikes for medium trips on flat ground or downhill, and e-bikes for longer uphill trips. However, Zou et al. [36] found that high-traffic roads are the most popular facilities for shared e-kick scooters because they have cycle paths. McKenzie [31] pointed out that e-PMVs services offer faster journeys during peak hours, while in central areas of the city, the times are almost like ride-hailing services. Finally, Nocerino et al. [32] showed that e-kick scooters are effective for letter and small package deliveries, not bulky boxes.

4.1.2. Modal Share

Most studies agree that e-PMVs cover short distances (i.e., within 5 km) and help reach other stops and/or stations. Nevertheless, a controversial issue is what transport modes they are replacing.

Table 3 summarizes a few studies concerning the modal shift towards e-PMVs.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--|---|---------------|---|--|---|---|--|
| Authors (lear) | Location, City | Type of Study | Data Sources | Sample Size | | Analytical 1001 | 0 |
| Almannaa et al. 2020 [16] | USA, Austin | Quantitative | General automatic data | 6 million trips | 3 December 2018–20 May 2019 | Cluster analysis | Compare the trip pattern between e-kick scooters and dockless e-bike |
| Bai & Jiao (2020) [17] | USA, Austin, Minneapolis | Quantitative | General automatic data | 905 trips (Austin) 619 trips (Minneapolis) | n/a | GIS method and negative binomial regression models | Analyze the space-time patterns and the effects of the land use factors on e-kick scooters ridership Examine trips patterns of e-kick |
| Caspi, Smart & Noland (2020) [18] | USA, Austin | Quantitative | General automatic data | 2,237,588 dockless trips | 15 August 2018–28 February 2019 | Descriptive statistics and spatial regression models | scooters and the effect of the built environment, land use, and demographics on trip generation |
| Davies, Blazejewski, & Sherriff (2020) [19] | n/a | Descriptive | n/a | n/a | n/a | Qualitative description | Show how e-PMVs can be used for tourism |
| Espinoza et al. (2019) [20] | USA, Atlanta | Quantitative | API automatic data | n/a | 26 January–1 February 2019; 2–5, 10–13, 15–19, February 2019; 26 February–5 March 2019 | GIS method and cluster analysis | Examine the role of e-kick scooters play in the mobility space (e.g., the purpose of each e-kick scooters trip) |
| Hardt & Bogenberger (2017) [21] | Germany, Munich | Qualitative | Travel diaries | 38 participants | May–July 2016 | Synthesis and codification of the travel diaries text | Understand the use, field of applications, and constraints of e-kick scooters |
| Hardt & Bogenberger (2019) [22] | Germany, Munich | Qualitative | Travel diaries and pre-post survey | 38 participants | 56 days | Synthesis and codification of the travel diaries text and pre-post survey | Understand the use, field of applications, and constraints of e-kick scooters with advantage and disadvantages Examine the temporal, land use, |
| Hawa et al. (2020) [23] | USA, Washington DC | Quantitative | API automatic data | 240,624 locations | 12–14 May; 16 May; 1 June; 14 June 2019 | Four multi-level mixed effects regression models | transport infrastructure, and weather factors that influence the e-kick scooters presence and their variations throughout the day |
| Jiao & Bai (2020) [24] | USA, Austin | Quantitative | General automatic data | 1,74 million trips | April 2018–February 2019 | GIS and negative binomial regression models | Analyze the space-time patterns and the effects of the land use factors on e-kick scooters ridership |
| Krizek & McGuckin (2019) [25] | USA | Descriptive | Survey | 9363 trips | 2017 | Qualitative description | Study who and for what kind of trips e-PMVs are used |
| Markvica, Schwieger & Aleksa (2020) [26] | Austria, Vienna | Descriptive | APP automatic data, online survey, course exercises | 51 participants, 128 respondents, 94 pupils | September 2018–June 2019; May–July 2019; two days in June 2019 | 'Triangulation' research strategy | Investigate the potential of e-kick scooters as last-mile options and their spatial and infrastructural implications |
| Mathew et al. (2019) [27] | USA, Indianapolis | Quantitative | General automatic data | 425,000 trips | 4 September–30 November 2018 | Descriptive statistics | Analyze the space-time patterns of e-kick scooters |
| Mathew, Liu & Bullock (2019) [28] | USA, Indianapolis | Quantitative | General automatic data | 532,190 trips | 4 September 2018–28 February 2019 | Negative Binomial Model | Examine the weather impact on urban e-kick scooters utilization |
| McKenzie (2019) [29] | Canada, Montreal Germany, Berlin USA, Los Angeles | Descriptive | API automatic data | 547,069 trips | 60 days | Qualitative description | Examine space-time mobility trips data of e-PMVs to assess the similarity among cities |
| McKenzie (2019) [30] | USA, Washington DC | Quantitative | API automatic data (e-scooters), Open data (bike) | 1,005,788 trips | 13 June-23 October 2018 | Descriptive statistics | Identify differences and similarities between dockless e-kick scooters and bike-sharing services |
| McKenzie (2020) [31] | USA, Washington DC | Quantitative | API automatic data (e-scooters), Open data (ride-hailing) | 6 mobility services | December 2018–March 2019 | Descriptive statistics | Identify space-time differences and similarities between e-PMVs services and between e-PMVs and ride-hailing services. |

Table 2. Trip pattern studies with e-PMVs.

Table 2. Cont.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|-----------------------------|--|---------------|------------------------|---|-----------------------------------|---|--|
| Nocerino et al. (2016) [32] | Croatia, Italy, Netherland, Portugal, Slovenia, Spain, Sweden | Descriptive | General automatic data | 39 companies test 74 electric vehicles | 3–12 months | Qualitative description | Study the potential of e-bikes and e-kick scooters for goods delivery in urban areas |
| Noland (2019) [33] | USA, Louisville | Quantitative | General automatic data | 88,042 records | 8 August 2018–28 February 2019 | Regression models | Show space-time data on shared e-kick scooters trip patterns |
| Reck et al. (2020) [34] | Switzerland, Zurich | Quantitative | General automatic data | 46,000 trips of 5 shared e-PMVs providers | 8 January–23 January 2020 | Bivariate relationships and a Multinomial Logit Model (MNL) | Compare bike, e-bike, and e-kick scooters usage patterns |
| Tuncer & Brown (2020) [35] | France, Paris | Qualitative | Personal interviews | 30 interviews | 5 weeks of observation | Synthesis and coding of interview text | Examine how to move with an e-kick scooter and coordination with other road users |
| Zou et al. (2020) [36] | USA, Washington DC | Quantitative | API automatic data | 138,362 records | 11 March–14 April 2019 | Descriptive statistics and spatial analysis | Analyze travel patterns and trajectories to understand the interaction with road design and vehicular traffic |

Table 3. Studies referring to the e-PMVs' transport impact.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--------------------------------------|----------------------------|---------------|--|--|-------------------|--|--|
| Berge (2019) [37] | Norway, Oslo | Quantitative | Interviews | 431 participants | Summer 2018 | Descriptive statistics | Examine the modes of transport replaced by the e-kick scooters |
| Hyvönen, Repo & Lammi (2016) [38] | Finland | Quantitative | Online survey | 1030 participants | n/a | Descriptive statistics | Examine how e-kick scooters could replace current transportation systems |
| Lee et al. (2019) [39] | USA, New York, Portland | Quantitative | General automatic data, National survey | 700,000 trips | 120 days | Log-log regression model and nonlinear multifactor model | Analyze the prediction of e-kick scooters demand and modal substitution of e-kick scooters |
| Tomita et al. (2016) [40] | Japan, Tsukuba | Quantitative | General automatic data | 4 sharing stations; 4 personal mobility devices; 60 registered users | September 2014 | Simulation multi-agent model | Analyze the modal shift due to the introduction of Segway |

Tomita et al. [40] showed that, for the sharing system, 8% of potential users said they would use e-PMVs if there were enough availability of them. Furthermore, the possibility of using them to reach train stations did not strongly influence the share of potential users.

Hyvönen, Repo & Lammi [38] and Berge [37] agreed that consumers are interested in trying light electric vehicles, but most of them would use them as a substitute for regular cycling and walking, while only 1 out of 4 would replace public transport. Conversely, Lee et al. [39], calibrating the demand forecasting model on Portland data and then applying it to Manhattan, showed that 1% of taxi journeys to access/exit public transport was replaced by e-kick scooters.

Although no journal article provided exhaustive information on what transport modes e-PMVs are replacing, the EIT Urban Mobility Report [41] shows an interesting comparison between some US and European reports on modal share. Figure 4 shows how the shift is different according to each city and that e-kick scooters mainly replace walking except for San Francisco, where they are replacing taxis. However, more research is still needed to provide strong general conclusions.



Figure 4. E-kick scooters and their impact on the modal share. Source: Authors' elaboration based on EIT Urban Mobility Report [41].

4.1.3. User Profiling

Some studies contributed to profiling e-kick scooters' users by investigating sociodemographic and travel characteristics (Table 4). User profiling was examined according to the "one-size-fits-all," a priori, and a posteriori (or post hoc) segmentation.

The "one-size-fits-all" segmentation outlined a representative e-kick scooters user. By multiple and logistic regression analysis, Huang & Lin [45] showed that age and gender differences affect the purchase and e-kick scooter use. In buying an e-kick scooter, men consider convenience, women consider the price, while under-20 consider the appearance. As for the reasons for using the e-kick scooter, respondents aged between 40 and 49 years showed a higher percentage for commuting purposes than younger respondents, and women showed a higher percentage in sport and leisure purposes than men. Furthermore, females felt stronger negative emotions than men. Otherwise, Fitt & Curl [44] defined that the most mentioned groups of e-kick scooter users were young people (118) and commuters and businesspeople (71).

Using a priori segmentation, possible users' characteristics on predefined segments have been identified and defined. More precisely, the literature aimed to learn more about the demographics and motivations of a specific category of e-kick scooters users. Using this approach, Eccarius & Lu [43] and Sanders, Brainon-Calles & Nelson [46] investigated a sample of university students and a sample of university staff and analyzed the frequency of use of the sharing service, respectively. Eccarius & Lu [43] showed that the compatibility of shared e-kick scooters with transport needs is the most relevant factor for the intention of use, but also awareness-knowledge and environmental values played an important role in the formation of intentions. Sanders, Brainon-Calles & Nelson [46] identified men with upper-middle-income (between \$ 50,000 and 75,000 dollars), between 25 and 34 years old, as the main users. Furthermore, there are no significant differences regarding ethnicity.

By post hoc segmentation, Degele et al. [42] aimed to discover groups e-kick scootersoriented by identifying specific user characteristics. Some clustering models and procedures have been adopted to identify segments within the data. They identified four main categories of users: expert (frequent users), occasional divided by age (over and under 40), and casual. Expert users are a small share of customers (4%) with an average age of 34, who travel mainly on Wednesdays in a distance of about 5.7 km and have an average time between rides of 4.6 days. Occasional users over 40 years have an average age of 48, represent approximately 24% of customers, use e-kick scooters more commonly on Fridays, and have an average time between rides of 25 days. Occasional users under 40 have an average age of 28, represent 58% of the market, use e-kick scooters more commonly on Saturdays, and have an average time between rides of 19.5 days. Finally, casual users are 14% of customers, with an average age of 35, who use e-kick scooters most commonly on Saturdays and have an average time between rides of 105.7 days.

4.1.4. Driving and Parking Behavior

Some papers studied the behavior of e-PMVs users when driving or parking (Table 5). These behaviors affect the relationship between e-PMVs users and other road users with reference to their circulation in the public space and the places where they park. The literature highlights as primary factors of driver behavior the speed, road infrastructure characteristics, and user types.

Concerning driving behavior, Arellano & Fang [47] showed that speed varies by infrastructures type (e.g., streets, sidewalks, and a mixed-use path) and gender. Indeed, e-kick scooters users' travel faster on roads and slower on pavements and mixed-use routes, move slower than cyclists and slow down encounter pedestrians. Furthermore, male users travel faster than females. Finally, e-kick scooter users are less distracted by cell phones, but they use headphones (16%) and few wear helmets (2%). Tuncer et al. [52] highlighted that to reach destinations faster and/or without being stopped, e-PMVs users quickly turn into pedestrians (getting off and on the device), "playing" with traffic rules. By getting off the device and walking, they can join pedestrians on sidewalks or cross with a red light, which gives them the right to keep moving. Furthermore, they try to blend in both in the public space in general and in encounters with pedestrians. Nishiuchi, Shiomi, & Todoroki [51] analyzed the driving behavior of experienced users and not on the phases of acceleration, deceleration, slalom, pedestrian overtaking, and emergency braking over a public road. Experienced users decelerate more gently, move faster during slalom, and drive in an agile and fast manner when they encounter pedestrians. Conversely, there is not much difference in emergency braking. Moreover, there is a similarity to the braking behavior of a bicycle.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|---|--|---------------|---------------------------|-----------------|-----------------------------|---|---|
| Degele et al. (2018) [42] | Germany | Quantitative | General automatic data | 53,000 data | 22 April–20 October 2017 | Cluster analysis | Analyze the behavioral and demographic segmentation of shared e-kick scooters |
| Eccarius & Lu (2020) [43] | Taiwan | Quantitative | Survey | 425 respondents | n/a | Factor analysis and structural equation modelling | Investigate the factors influencing college students' intention to use shared e-kick scooters |
| Fitt & Curl (2020)[44] | New Zealand, Auckland, Hutt Valley, Christchurch, Dunedin | Quantitative | Online survey | 491 respondents | February–March 2019 | Descriptive statistics | Indicate the characteristics of e-kick scooters users |
| Huang & Lin (2019) [45] | Taiwan | Quantitative | Online survey | 190 individuals | n/a | One-way analysis of variance, Multiple and Logistic Regression Analysis | Understand the age and gender differences that influence the use of e-kick scooters |
| Sanders, Branion-Calles & Nelson (2020) [46] | USA, Tempe | Quantitative | Survey | 1256 responses | 2 May 2019 | Descriptive statistics | Indicate the socio-demographic characteristics and travel patterns of e-kick scooters users |

Table 4. User-profiling studies with e-PMVs.

Table 5. Studies on driving and parking behavior.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--|---|---------------|---------------------------|--|-------------------------------|---|--|
| Arellano & Fang (2019) [47] | USA, San Jose | Quantitative | Observation | 330 e-kick scooter riders | October 2018–February 2019 | Descriptive statistics | Observe the behavior of e-kick scooter drivers and how they compare with pedestrians and cyclists. |
| | USA, Austin, Portland, San | | | | | | I |
| Brown et al. (2020) [48] | Francisco, Santa Monica, Washington DC | Quantitative | Observation | 3666 data | July–August 2019 | Descriptive statistics | Examine parking practices |
| Fang et al. (2018) [49] | USA, San Jose | Descriptive | Observation | 530 shared e-kick scooters 606 parked e-kick scooter, | June–July 2018 | Qualitative description | Examine parking practices Analyze how shared e-kick scooters are |
| James et al. (2019) [50] | USA, Rosslyn | Quantitative | Observation, survey | 181 e-kick scooters riders and non-riders | 4 April–24 April 2019 | Descriptive statistics | parked and pedestrians' perceptions of vehicle safety |
| Nishiuchi, Shiomi, & Todoroki (2015) [51] | Japan | Quantitative | General automatic data | 14 subjects | n/a | Two-way analysis of variance | Analyze user behavior in 5 driving phases |
| Tuncer et al. (2020) [52] | France, Paris | Descriptive | Video observation | 3 e-kick scooter riders | 2018 | Ethnomethodology and multimodal conversation analysis | Examine the driving practices of e-kick scooter users and their interactions with pedestrians |

As for parking behavior, it is observed whether e-PMVs users park (mainly e-kick scooters) in the appropriate areas or if they hinder the use of the infrastructure for other road users. Brown et al. [48], Fang et al. [49], and James et al. [50], agreed that even fewer parked e-kick scooters hinder traffic. The problems encountered so far in the car park concern possible access blocks for the disabled (2%, [49]) or to pedestrians (4%–10%, [50]), reducing the passage to less than 80 cm [48]. Further issues relate to comfort and perceived safety between users and non-users. According to James et al. [50], 76% of non-users and 24% of users reported feeling insecure while walking around dockless e-kick scooters.

4.1.5. Regulation and Organization of the Urban Spaces

Some studies addressed issues related to the regulation and organization of urban spaces (Table 6). These emerging vehicles constitute an additional transport system to already existing ones, occupy additional urban space (which is already diminishing) and, thus, need specific regulations. They also need additional infrastructures to be operated (e.g., charging points, parking slots).

To account for these issues, several authors investigated some facets.

For what concerns the providers of the e-kick scooter sharing system, Janssen et al. [58] showed that there are several common policies among cities. The number of providers is not fixed but has a maximum limit, each operator must pay a registration fee and a permit, and they have areas in which to circulate e-kick scooters. Also, not all e-kick scooters providers use the data-sharing platform. In some cities, there are restrictions on timetables, and on the reserved areas where e-kick scooters are or not allowed to park, but all cities reserve the right to remove improperly parked e-kick scooters.

As for the organization of urban spaces, some studies show how e-PMVs can create conflicts during traffic and how their diffusion increases the pressure on infrastructures. Regarding conflicts, Gössling [57] showed that e-PMVs create struggles in areas where they operate, with a difference in speeds and safety. Cases of irresponsible riders' behavior, disorder, and vandalism are frequent, especially in large cities. Instead, Jiménez, De La Fuente, & Hernández-Galán [59] showed that the integration between pedestrians and e-PMVs in the same urban space had not taken place yet. Therefore, they suggest different strategies to ensure urban accessibility, such as user regulation (i.e., training considers the functional diversity of pedestrians) and regulation for the use of devices (i.e., location of use and maximum speed). Conversely, to improve the road safety of coexistence with other pedestrians, they recommend ease of detection (e.g., using easily identifiable colors or inclusion of sound or light mechanisms) and consider the diversity in body size and reaction rates. On the other hand, as for the pressure on existing infrastructures, Butrina et al. [53] showed how several municipalities are adapting to changing sidewalk pressures because overcrowding on pavements spills into lanes, particularly acute impacts public road safety. Public administrations are analyzing data in real-time to be able to manage the flooring dynamically.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample size | Period Covered | Analytical Tool | Relevant Insights |
|---|--|---------------|---|---|--|---|--|
| Butrina et al. (2020) [53] | USA | Qualitative | Interviews | One/two participants from the 10-interviewee city | May–July 2019 | Synthesis and coding of interview text | Show how municipalities are adapting to the new pressures owing to e-PMVs services on their sidewalk |
| Chen et al. (2018) [54] | n/a | Quantitative | n/a | n/a | n/a | Mixed-Integer Linear Programming model and Particle swarm optimization algorithm | Identify the location of charging stations |
| Clark, Atkinson- Palombo & Garrick (2019) [55] | n/a | Descriptive | n/a | n/a | n/a | Qualitative description | Analyze the social acceptance of Segway |
| Fang, Chang & Yu (2014) [56] | Taiwan, Penghu Island | Quantitative | General automatic data | 27 locations, 12,132 transactions | n/a | Quantitative statistical data processing and analysis methods | Identify the characteristics of users' charging behaviors, the optimal placement of charging stations, the users' charging time, and usage frequency. |
| Gössling (2020) [57] | Austria, Australia, Denmark, France, New Zealand, Spain, Sweden, Switzerland, USA | Descriptive | Local media report | 173 news items | December 2017–August 2019 Depends on the cities | Qualitative description | Analyze the problems associated with the introduction of e-kick scooters and the different sharing operators |
| Janssen et al. (2020) [58] | USA, Austin, Charlotte, Denver, Indianapolis, Louisville, Memphis, Minneapolis, Nashville, Raleigh, Seattle | Descriptive | Multiple sources (e.g., police document, statistical data) | 10 mid-sized peer cities | December 2018–June 2019 | Qualitative description | Examine how scooter policies compare between cities and over time. |
| Jiménez, De La Fuente, & Hernández-Galán (2018) [59] | n/a | Descriptive | n/a | n/a | n/a | Qualitative description | Analyze the coexistence and accessibility problems between pedestrians and e-PMVs |
| Lo et al. (2020) [60] | New Zealand | Quantitative | Online survey | 230 responses | May 2019 | Descriptive statistics | Study the impact of legislation on the widespread use of e-kick scooters |
| Moran, Laa & Emberger (2020) [61] | Austria, Vienna | Quantitative | Automatic data manually digitized | 6 e-kick scooters operators | June-August 2019 | Spatial analysis | Analyze the spatial variance in e-kick scooter geofences |

Table 6. Studies on regulations and organization of the urban spaces.

Moran, Laa & Emberger [61] focused on the spatial variation of geographic barriers of e-kick scooters and how these differences relate to existing regulations. The six providers analyzed have overlapping virtual fences and no-parking zones defined within the city but of different sizes.

Since e-PMVs need battery recharging, a handful of studies investigated some issues related to the charging points. Chen et al. [54] showed that the optimal location for e-PMVs charging systems is in the function of the greatest number of times a station is used. Moreover, the optimal location is affected by the minimum total cost and the maximum capacity of the service (i.e., the maximum number of rechargeable vehicles in a day), the land-pricing, the service distance, and the installed capacity. In addition, Fang, Chang & Yu [56] showed that occasional users recharge their e-PMVs for less than 2 h, while "repeated" and "high frequency" users (i.e., students or workers) occupy the recharging station more: about 4/5 h. It might be because the e-PMVs stay parked for more time, and the parking is adopted as a depot.

The regulation and the organization of the urban spaces originate challenges regarding the acceptance of e-PMVs among the users. Social acceptance of e-PMVs means the achievement of knowledge of the means of transport from different viewpoints (e.g., design, road safety, circulation). This level differs between non-users and experienced users, who are more likely to accept e-PMVs.

Clark, Atkinson-Palombo & Garrick [55] showed that high prices, legislative and spatial issues, and a lack of appeal to consumers presented challenges to e-PMVs acceptance. Therefore, their social, economic, and environmental costs and regulatory issues presented barriers to their diffusion. Lo et al. [60] confirmed the legislation could impact the widespread use of shared e-kick scooters. Frequent users (traveling more than 3 times a week) are strongly opposed to any regulation, while non-users would be more inclined to try e-kick scooters if there were mandatory helmet rules.

4.2. Exogenous Issues: Impact on Road Safety and Environment

Forty-three studies examined the impact of e-PMVs on road safety and the environment. They analyze the main types of injuries and most affected people (crashes severity), the road safety implications related to conflicts between e-PMVs and pedestrians, the use of user protection devices, and new technological systems for safe driving and a miscellaneous regarding the types of crashes. Furthermore, a handful of studies examine whether e-PMVs can be considered a sustainable means of transport and their environmental impact.

4.2.1. Crashes Severity

Many studies addressed the road safety of e-PMVs, considering the types and severity of injuries for both e-PMVs' drivers and the other road users involved (Table 7). Most of the studies focused on the types of injuries, including patient demographics, the type of crashes, and compliance with traffic rules (e.g., the use of personal protective equipment, a good psychophysical state of the driver). Conversely, a handful of studies focused on the historical trends of hospitalizations in terms of the number of patients, seasons, months, or days of the week. All these studies adopted crash data gathered from the emergency department visits, which are quite detailed from a medical perspective and include data on slight and minor injuries.

According to a general overview, the most reported injuries are neurological, maxillofacial, orthopedic, and thoracic types. They involve the upper and lower part of the body and head representing the more exposed and vulnerable parts of the overall body of an e-PMVs' user [63,75,79]. More precisely, details of type of injuries are shown in Figure 5.



Figure 5. Type of injuries.

Further studies also correlated patient demographics such as age and gender to injury data. All these studies showed that patients are mostly males [78]. Moreover, patients involved in a crash are young aged between 18 and 25 [65] and middle-aged from 34 and 38 [64,66,70,73,74,81,83].

Other studies also specified whether the patient was the e-PMVs' driver or a road user, and the crash type. While confirming the results of type and severities of injuries and the age, they showed that 97% of patient were e-PMVs drivers [69]; they injured themselves by losing balance while driving (81%), were hit by an object (3%) [71,72], and 16% were hit by a car [67]. Several studies argued that injuries were reported due to incorrect driving behavior, such as driving under the influence of alcohol or drugs and considering violations to road regulation code. Puzio et al. [80] showed that no one of the hospitalized patients wore a helmet, and about 34% drove under the influence of alcohol or drugs during the crash. Following this type of misguided, Haworth & Schramm [68], Badeau et al. [62], and Mitchell et al. [76] crossed data on injuries and incorrect driver behavior and showed that the main types of injuries reported fall within those previously analyzed. Even Kobayshi et al. [73] and Bekhit et al. [64] confirmed it.

Störmann et al. [82] enlarged previous studies' results, grouping the main types of patient injuries, demographics, type of crashes, improper driving, and confirming the findings. They showed that crashes are greatest between August and September, many of them on weekends. While confirming the admission on the weekends, Vernon et al. [84] added the number of hospitalizations increased between April and July. In addition, Namiri et al. [77] showed that victims aged between 18 and 34 increased by 354% between 2014 and 2018.

4.2.2. Conflicts between e-PMVs and Pedestrian

Some studies investigated how pedestrians perceive e-PMVs and the interactions between them and congested pedestrian areas (Table 8). Additionally, they identified areas where there is a greater chance of encountering e-PMVs.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|---------------------------------------|-----------------------|---------------|---|-------------------------------------|--|---|--|
| Badeau et al. (2019) [62] | USA, Salt Lake City | Quantitative | Emergency department visits | 58 patients | June 15–15 November 2017 and 2018 | Descriptive statistics | Identify the type of injury |
| Beck et al. (2020) [63] | New Zealand, Dunedin | Quantitative | Emergency department visits | 56 patients | 6 weeks January 10–20 February 2018 and 2019 | Descriptive statistics | Describe the injury patterns associated with e-kick scooters |
| Bekhit et al. (2020) [64] | New Zealand, Auckland | Quantitative | Emergency department visits | 770 patients | 20 September 2018–20 April 2019 | Descriptive statistics | Show the main types of injuries, patient demographics, and incorrect driving |
| Blomberg et al. (2019) [65] | Denmark, Copenhagen | Quantitative | Emergency department visits | 468 patients | January 2016–July 2019 | Descriptive statistics | Describe injuries and patient demographics related to e-kick scooter use |
| Bloom et al. (2020) [66] | USA, Los Angeles | Quantitative | Emergency department visits | 248 patients | 1 February–1 December 2018 | Descriptive statistics | Study the injuries related to e-kick scooters use |
| Dhillon et al. (2020) [67] | USA, California | Quantitative | Emergency department visits | 87 patients | January-December 2018 | Descriptive statistics | Show the variation in hospital admissions and outcomes for injuries related to e-kick scooters use |
| Haworth & Schramm (2019) [68] | Australia, Brisbane | Quantitative | Emergency department visits | 785 e-kick scooters 109 patients | Two months in early 2019 | Descriptive statistics | Track the number of e-kick scooters involved in crashes and the types of injuries |
| Ishmael et al. (2020) [69] | USA, Los Angeles | Quantitative | Emergency department visits | 73 patients | September 2017–August 2019 | Descriptive statistics | Show operational orthopedic injuries related to e-kick scooter crashes |
| Islam et al. (2020) [70] | Canada, Calgary | Quantitative | Emergency department visits | 33 patients | 8 July-30 September 2019 | Descriptive statistics | Analyze data of injury related to the use of e-kick scooters |
| Kim et al. (2018) [71] | Korea, Incheon | Quantitative | Emergency department visits | 65 patients | January 2016–December 2017 | Descriptive statistics and logistic regression | Show injury types, patient demographics, and crashes dynamics |
| King et al. (2020) [72] | Singapore, Singapore | Quantitative | Emergency department visits | 259 patients | 1 January 2016–31 December 2016 | Descriptive statistics | Analyze injury patterns related to the use of e-kick scooters |
| Kobayshi et al. (2019) [73] | USA, San Diego | Quantitative | Emergency department visits | 103 patients | 1 September 2017–31 October 2018 | Descriptive statistics | Show the main types of injuries, patient demographics, and incorrect driving |
| Liew, Wee & Pek (2020) [74] | Singapore, Singapore | Quantitative | Emergency department visits | 36 patients | From 2015 to 2016 | Descriptive statistics | Characterize the severity of e-kick scooter-related injuries |
| Mayhew & Berging (2019) [75] | New Zealand, Auckland | Quantitative | Emergency department visits | 64 patients | August 15–15 December 2018 | Descriptive statistics | Quantify the severity of injuries caused by an e-kick scooter |
| Mitchell et al. (2019) [76] | Australia, Brisbane | Quantitative | Emergency department visits | 54 patients | 23 November 2018–23 January 2019 | Descriptive statistics | Analyze the type of incorrect driving and crashes |
| Namiri et al. (2020) [77] | USA | Quantitative | National Emergency department visits | 1037 injuries | 2014–2018 | Descriptive statistics and linear regression | Study the trends in injuries and hospital admissions |
| Nellamattathil & Amber (2020) [78] | USA, Washington DC | Descriptive | Radiology report database | 54 patients | 1 September 2017–1 December 2018 | Qualitative description | Identify the pattern of injuries on diagnostic imaging Examine the literature review to |
| Pourmand et al. (2018) [79] | USA, Austria, Denmark | Descriptive | Scientific databases (PubMed) | 6 studies 135 patients | January 1990–May 2017 | Qualitative description | understand the types of injuries associated with Segway use, patient demographics, the context of injuries, and the cost associated with injuries |
| Puzio et al. (2020) [80] | USA, Indianapolis | Quantitative | Emergency department visits | 92 patients | 4 September–4 November 2018 | Descriptive statistics | Characterize the epidemiology of injuries |
| Roider et al. (2016) [81] | Austria, Vienna | Quantitative | Emergency department visits | 86 patients | January 2010–December 2012 | Descriptive statistics | Analyze the injuries associated with the use of the Segway for sightseeing |

Table 7. Studies on crashes severity.

| | Table 7. Cont. | | | | | | | | | | |
|-----------------------------|--------------------|---------------|--------------------------------|--------------|------------------------------------|------------------------|---|--|--|--|--|
| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights | | | | |
| Störmann et al. (2020) [82] | Germany, Frankfurt | Quantitative | Emergency department visits | 76 patients | July 2019–March 2020 | Descriptive statistics | Identify injury patterns, patient demographics, type of accident, type of driving and growth in hospitalizations | | | | |
| Trivedi et al. (2019) [83] | USA, Los Angeles | Quantitative | Emergency department visits | 249 patients | 1 September 2017–31 August 2018 | Descriptive statistics | Characterize injuries related to the use of e-kick scooters | | | | |
| Vernon et al. (2020) [84] | USA, Atlanta | Quantitative | Emergency department visits | 293 patients | 3 May 2018–15 August 2019 | Descriptive statistics | Assess the health care impact of e-kick scooter crashes. | | | | |

2019

Table 8. Studies on conflicts between e-PMVs and pedestrian.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--------------------------------|----------------------|---------------|---|--------------------------------|--------------------------------|---|--|
| Che, Lum & Wong (2020) [85] | Singapore, Singapore | Quantitative | Observation (Head-mounted displays) | 60 people 12 scenarios | n/a | Descriptive statistics | Evaluate the safety perceived degree considering speed, even when overtaking between e-kick scooters and pedestrians |
| Hasegawa et al. (2018) [86] | n/a | Quantitative | General automatic data | 32 participants | February–March 2017 | Model based on social forced including acceleration and speed data | Evaluate the safety perceived degree considering the movement direction between Segway and pedestrians |
| Kuo et al. (2019) [87] | Singapore, Singapore | Quantitative | Online survey | 303 responses | December 2018–February 2019 | Descriptive statistics and structural equation model | Evaluate the pedestrians' levels of acceptance of e-PMVs, based on the intention to use a PMV, ease of use, usefulness, the perceived risk from PMV riders, and the environment. |
| Kuo et al. (2019) [88] | Singapore, Singapore | Quantitative | General automatic data, survey | 4 types of PMV 39 participants | December 2018–March 2019 | Descriptive statistics and machine learning algorithm | Understand the reaction of the pedestrians to the e-kick scooters, considering speed and gender |
| Maiti et al. (2019) [89] | USA, San Antonio | Quantitative | Smartwatch automatic data, post-study survey | 77 participants | Three months | Descriptive statistics | Investigate the safety issues due to e-kicl scooters services from the pedestrian's |
| Pham (2019) [90] | n/a | Theoretical | n/a | n/a | n/a | Simple microscopic model | perspective Propose an assistance system for study the interaction between an e-PMVs and pedestrians, considering the personal space |
| Pham et al. (2015) [91] | n/a | Theoretical | n/a | n/a | n/a | Refined microscopic model | Propose an assistance system for study the interaction between an e-PMVs and pedestrians, considering the personal space |

21 of 34

On pedestrians' perception of e-PMVs in shared paths, assessments were made on the degree of perceived safety when the vehicle moves nearby, considering the speed of the e-PMVs, the pedestrians' gender, and the degree of acceptance e-PMVs on shared paths. The acceptance phase can be defined as completed when the new device no longer represents a novelty but a familiar vehicle that has been in circulation for some time (e.g., bicycles). Che, Lum & Wong [85] suggested that e-PMVs speeds between 10 and 15 km/h are considered safe when e-PMVs overtake a pedestrian, while a speed up to 15 km/h is safe when they are face-to-face with pedestrians. Indeed, pedestrians feel uncomfortable when a device approaches them faster than 15 km/h, and females usually feel more uncomfortable than males [88].

The results differ in the acceptance of e-PMVs by pedestrians. Kuo et al. [87] showed how this acceptance is more influenced by context than individual behavior. An e-PMV is less "dangerous" for pedestrians in relation to the space in which they circulate, rather than about the speed used or the direction of the device. Furthermore, there are also differences between experienced users and non-users.

Hasegawa et al. [86] showed that the subjective perception of danger depends on the movement direction in which both pedestrians and e-PMVs approach each other. Pedestrians consider e-PMVs to be more dangerous when they arrive from the front rather than from behind. In addition, when e-PMVs move in a pedestrian flow, they may generate a negative psychological effect. People can feel fear and discomfort when something invades their personal space, which induces psychological stress [92]. This effect was studied by Pham [90] and Pham et al. [91], who showed the levels of discomfort and fear of e-PMVs increase as the pedestrian density increases. The highest density of encounters between e-PMVs and pedestrians was recorded around campus residences, parking lots and off-campus apartment complexes during lessons and study hours [89]. Therefore, these zones are potentially the most dangerous to pedestrians.

4.2.3. Safe Driving

Some studies showed several elements that could contribute to safe driving (Table 9). In this sense, the researchers analyzed the social networks of some sharing companies to understand how much they emphasize the use of protective devices. Moreover, they showed new technologies developed to assess whether users are driving incorrectly (e.g., carrying people or things, driving drunk, etc.) and issues related to the design of the device itself. An in-depth analysis examined the dynamics of crashes, while others focused on the safety distance to be kept when overtaking.

Consumer behavior can generally be influenced by how companies promote and demonstrate their products on popular social media platforms, such as Instagram and Twitter. In fact, Allem & Majmundar [93] and Dormanesh, Majmundar & Allem [94] showed that only about 10% of Bird company's posts show users wearing protective clothing, while Tier Mobility about 26%.

The growing popularity of e-kick scooters has led to the rising of individual safety issues associated with unsafe driving behavior. Terrell [98] proposed a new technology: mounting a weight detection mechanism on an e-kick scooter that enables determining its driving behavior and indicates an unsafe and/or not permitted use. Identified driving patterns can indicate whether users are driving under the influence of alcohol or drugs, carrying passengers, driving recklessly, etc. Conversely, Kim et al. [96] examined the risks associated with electric scooter users by analyzing the influence of face direction while driving via a smartphone application. They showed that the greater the angle between the direction of the scooter and the direction of the gaze (face angle), the greater the deviation of the traveler in the direction opposite to that observed by the subject.

In terms of public safety, even vehicle design and overtaking maneuvers are relevant. Nisson, Ley & Chu [97] pointed out that e-kick scooters, being silent, dark-colored, and often light-free vehicles, are dangerous vehicles for drivers and road users. In addition, since collisions during overtaking maneuvers are defined as one of the main causes of fatal crashes on two wheels [99], the minimum distance that two-wheeled vehicles (bicycles, e-bikes, and e-kick scooters) should use to overtake a car should be 1.54 m [95].

4.2.4. Miscellaneous on Road Safety

Three studies cannot be classified in the previous sections, so a new one has been created (Table 10). Two studies analyzed the kinematic process of crashes by evaluating head injuries and crashes patterns.

Xu et al. [100,101] have investigated head trauma as a type of injury for pedestrians and e-PMVs users. They showed no appreciable differences on impact to the ground (i.e., the severity of injuries is quite similar). Conversely, the risk of injury increases with the vehicle and the e-kick scooters speeds.

As for the dynamics of crashes, owing to the scarcity of data, it is difficult to understand the number of crashes with e-kick scooters involved. For this, Yang et al. [102] tried to describe the crash patterns related to e-kick scooters' use. Results pointed out that both children and the elderly are prone to serious injuries and that crashes are more likely to occur at night than during the day. Additionally, the outcome of the crash may be related to the gender difference, with women involved in multiple falls. The main types of a reported crash are collisions with a car and falling alone.

4.2.5. Environmental Impact

Few studies addressed the issue of the environmental impact of e-PMVs (Table 11), but all of them agreed that although the emissions while driving are equal to zero, the entire life cycle must be considered. The Life Cycle Assessment (LCA) is a quantitative environmental impact assessment method, which enables to calculate the environmental impacts of a product or service through the whole life cycle. The main goal of LCA is to calculate impacts, compare different products and/or services, and highlight improvement options.

Severengiz et al. [106] showed that the Global Warming Potential (GWP) associated with the use of shared e-kick scooters is dominated by the production phase, especially the production of aluminum parts. In addition to production, the lifespan, the distances to collect the batteries or scooters, the type of collection vehicle, and the mix of electricity for charging the scooters are important factors. Hollingsworth, Copeland & Johnson [104] also agreed that materials, manufacturing, and automotive use for collecting e-kick scooters and charging them dominate the impacts of global warming associated with the use of shared e-kick scooters.

However, De Bortoli & Christoforou [103] argued that it is necessary to combine the useful life kilometers of e-kick scooters and the environmental impacts of maintenance phases to make these devices sustainable. Moreau et al. [105] showed that dockless e-kick scooters must be used for at least 9.5 months to be an ecological solution for mobility in the current usage situation.

All contributors agree that the GWP results are sensitive to the e-kick scooters' duration over the years. In the current situation, the use of the e-kick scooter causes between 64 and 237 g of CO2-eq * $p \cdot km^{-1}$ and, after at least two years, this number can decrease between 20% and 30%. The value is visibly lower than the emissions from cars causing between 147 and 414 g of CO2-eq * $p \cdot km^{-1}$. It shows how modal choice is relevant for the environmental impact: the modal shift could be positive, as in the previous comparison with cars, or negative if an e-kick scooter replaces a sustainable transport mode with a lower LCA. For example, the use of an e-bike causes about 40 g of CO2-eq * $p \cdot km^{-1}$, and a bicycle causes 8 g of CO2-eq * $p \cdot km^{-1}$.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--|------------------|---------------|---------------------------------|--|--------------------------------------|-------------------------|--|
| Allem & Majmundar (2019) [93] | USA | Quantitative | Instagram posts | 324 posts | 22 September 2017–9 November 2018 | Descriptive statistics | Examine Bird's official Instagram account to determine how much it emphasizes safety devices in its posts |
| Dormanesh, Majmundar & Allem (2020) [94] | n/a | Quantitative | Instagram and Twitter posts | Instagram posts: Bird 287, Ties Mobility 190. Twitter posts: Bird 313, Ties Mobility 67 | 9 November 2018–7 October 2019 | Descriptive statistics | Examine Bird and Tier Mobility's official Instagram and Twitter accounts to determine how much they emphasize safety devices in their posts |
| Guo, Sayed & Zaki (2019) [95] | China, Kunming | Quantitative | Video observation | 352 overtaking | 1.30 h | Logit model | Examine the factors affecting the lateral distance when overtaking between two-wheeled devices and cars |
| Kim et al. (2018) [96] | Japan, Tsuruoka | Quantitative | App automatic data | 4 participants | n/a | Descriptive statistics | Evaluate the impact of the face angle on the trajectory while driving |
| Nisson, Ley & Chu (2020) [97] | USA, Los Angeles | Descriptive | n/a | n/a | n/a | Qualitative description | Provide recommendation on personal and public safety associated with e-kick scooter design. |
| Terrell (2019) [98] | n/a | Qualitative | Weight sensor automatic data | n/a | n/a | Qualitative description | Determine driving behavior that indicates unsafe and/or impermissible use |

Table 9. Studies on safe driving.

Table 10. Studies on miscellaneous on road safety.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|--------------------------|----------------|---------------|---------------|-------------|------------------------------------|--|--|
| Xu et al. (2016) [100] | n/a | Theoretical | n/a | n/a | n/a | Analytical model | Study the kinematic process in electric self-balancing scooters-vehicle crashes, evaluating only head injuries |
| Xu et al. (2016) [101] | n/a | Theoretical | n/a | n/a | n/a | Analytical model | Study the kinematic process in electric self-balancing scooters–vehicle crashes, evaluating head injuries during secondary contact with the ground evaluating |
| Yang et al. (2020) [102] | USA | Quantitative | Media reports | 169 crashes | 1 January 2017–31 December 2019 | Synthesis and coding of newspaper text | Describe the crashes patterns due to the use of e-kick scooters |

Table 11. Studies on the environmental impact.

| Authors (Year) | Location, City | Type of Study | Data Sources | Sample Size | Period Covered | Analytical Tool | Relevant Insights |
|---|--------------------|---------------|-----------------------------------|----------------|------------------|---------------------------------------|--|
| De Bortoli & Christoforou (2020) [103] | France, Paris | Quantitative | Survey data | 445 responses | Spring 2019 | Analytic model | Analyze the mathematical formalization of consequential Life Cycle Assessment of free-floating e-kick scooters |
| Hollingsworth, Copeland & Johnson (2019) [104] | USA, Raleigh | Quantitative | n/a | n/a | n/a | Monte Carlo analysis and scenarios | Analyze the Life Cycle Assessment of shared dockless e-kick scooters. |
| Moreau et al. (2020) [105] | Belgium, Bruxelles | Quantitative | General automatic data, Survey | 1181 responses | June-August 2019 | Descriptive statistics | Analyze the Life Cycle Assessment of dockles and personal e-kick scooters |
| Severengiz et al. (2020) [106] | Germany, Berlin | Quantitative | Survey with service providers | n/a | n/a | Descriptive statistics | Analyze the Life Cycle Assessment of shared e-kick scooters |

5. Towards the Development of Research on e-PMVs

Currently, research focused on the endogenous and exogenous issues affecting the massive spread of e-PMVs in urban contexts. It showed how these vehicles were integrated into the built environment. However, new studies are suggested for both endogenous and exogenous issues without any priority order.

Endogenous issues research reported valuable results of trip pattern characteristics (such as average speed, distance, day, time, etc.). In addition, some key factors affecting ridership were isolated at an aggregate level. However, most of these studies refer to USA cities. Therefore, it would be interesting to replicate the same analyses in European cities, which differ in territorial conformations, climate, regulations, and population habits. For instance, many European cities are older than the USA cites and present built environments that are more consolidated and restricted, which could affect the characteristics of e-PMVs. Despite their introduction being conceived to increase the sustainability of transport in cities, both in Europe and USA, only a handful of studies showed that their use has mainly replaced soft mobility rather than hard mobility (i.e., cars) by providing descriptive statistics. This does not enable for strong conclusions. Therefore, new studies and more refined statistical models would be recommended to further investigate this important issue.

Studies on user profiling are still few and new research to evaluate the propensity to use e-PMVs by means of probabilistic models seems crucial. Thus, it would be possible to understand what leads the user to use them, considering the emotional and non-rational side other than classical socio-demographic and travel behavior characteristics.

As for the travel behavior linked to public space use, further research is desirable regarding the characteristics and parking spaces and pavement suitable for the circulation of these new devices. As this survey showed, the literature lacks contributions to the effect of different pavement types where e-PMVs can circulate. In addition, some European countries issued specific regulations for the circulation of e-PMVs, equating them to bikes (or e-bike), thus providing indications regarding their circulation in urban areas both on cycling paths and traditional roads. However, the effect of these regulations requires attention because they present different characteristics. For instance, cyclists are seated on their vehicle, whereas the e-PMVs users must stand above the footboard. Moreover, bikes have large wheels and tires, which can generate a stabilizing gyroscopic effect and dissipates the shocks induced by the pavement irregularities. Conversely, e-PMVs are generally equipped with small diameter wheels, often made of a rigid material, which may not induce significant stabilizing and dissipative effects. Therefore, the similarity between bikes and e-PMVs could be questionable: the few studies that compared them usually refer to trip patterns, and the dynamic behavior of the vehicles is not considered. Only a few indications on the city's transport policy emerged from the literature, perhaps because e-PMVs are new vehicles and the regulations for their circulation are constantly updated. Hence, guidelines and policies are needed to integrate e-PMVs into public space and investigate the parameters that drive the location of recharging points (especially for shared vehicles) and the monitoring of e-PMVs as a sustainable mode of transport.

Research should also be deepened on exogenous issues related to road safety and the environment. This survey highlighted a lack of studies that should separate the crashes and traumas involving e-kick scooters to provide specific information about them. Indeed, there is no specific filing of injuries caused by e-kick scooters in hospitals, which is useful for a detailed analysis of patients, the type of injuries, and the number of injuries. It would be interesting to include these data in an accessible national database, useful to detect appropriate safety measures and recommendations.

As for pedestrian safety, future developments could focus on the effects and usefulness of signaling devices to be installed on devices to warn of the presence of important pedestrian flows. Furthermore, experimental studies should be conducted on personal protective equipment and dynamics and reconstructions of crashes. In this context, driver training before using e-kick scooters is essential, and the research contributions are too limited on this aspect. Even more so, e-PMVs can circulate in promiscuity with individual motor vehicles (e.g., cars). Therefore, an analysis of the risk of crashes encompassing statistical models of the probability of the occurrence, the severity of crashes, and the exposure variables would be an important research topic for classifying the paths where e-PMVs can be admitted, as already applied in public transport [107].

Finally, further research on the environmental impact is also desirable: it would be interesting to understand how ecological e-kick scooters are, especially in the construction and disposal phases, or in relation to the types of mobility they replace. There are still few (and mainly European) studies that show the importance of considering the entire life cycle of the device.

6. Conclusions

Recently, the diffusion of electric-powered Personal Mobility Vehicles (e-PMVs) in many worldwide cities led to several issues that have captured the attention of many scholars and practitioners. The scientific community has grown considerably over the past three years. Although there are many more studies (such as scientific literature in many national languages and professional reports), extensive search is done to analyze studies in English only. Hence, 90 publications have been revised to understand the several issues associated with the spread of e-PMVs. This survey classified the studies according to both endogenous issues (i.e., impact on transport and urban planning) and exogenous issues (i.e., impact on road safety and environment) and showed that research has evolved over the years to the increasing use of e-PMVs and the data availability.

Studies aimed at endogenous issues were mainly conducted in North America and Europe. That research was dedicated to defining travel patterns (i.e., analyzing the average length traveled, the time and speed of the journey), understanding where and when e-PMVs were used, including the diversity of ground and climatic conditions of the city. Further, the main reasons that encourage using these devices were reported, and which existing means of transport e-PMVs are replacing. This survey showed that e-PMVs are mostly used on weekends during the afternoon hours for leisure and free time trips. During the week, they may be used for commuting to/from work or to reach stops/stations. E-PMVs are most used in areas where there is a greater diversity of land use. The weather is an important factor in the choice of use of these devices. E-kick scooters are considered an attractive transportation mode for the feeling of freedom and fun and for reducing trip time. Also, it appears that e-PMVs will replace walking and cycling without reducing the use of private cars for short trips, but more studies are required to confirm previous results because few are identified.

Subsequently, the studies enable us to profile the user type through the segmentation "one-size-fit-all," a priori, and a posteriori. This literature showed that the main frequently users of e-PMVs are young men between 20 and 40 years old. Their behavior towards other road users is analyzed in the use of public space, during the driving and parking phases (i.e., if they obstruct circulation or if driving on shared spaces is appropriate), and some aspects related to the regulation and design of urban public space. The studies on user behavior showed that, while driving, users move slower than bicycles, slow down when pedestrians are present, and are agiler when they gain more riding experience. In addition, users often respect the rules of parking lots but "play" with traffic rules by getting on and off the devices to move like pedestrians. Nevertheless, only a small percentage of parked e-PMVs may create severe problems for the pedestrian flows. Conversely, studies showed that better effort is needed in speed regulation and traffic areas to avoid conflicts with other road users, which could also improve their acceptance. Finally, many issues that should be further addressed concern the charging points (e.g., frequency of use, costs, and capacity of the service).

Exogenous issues studies on road safety were mainly carried out in North America and Asia. They analyzed the degree of crashes severity based on their number and type and to the profiling of people subject to impact. The average age of injured people is between 30 and 35, mainly men who did not wear helmets. Many of them also drive under the influence of alcohol and/or drugs. The most reported types of injuries are head trauma, brain injury, and upper body injuries or fractures. Research also analyzed safety towards other road users (mainly pedestrians), referring to the perception that pedestrians have when they are in proximity to these devices while traveling.

Subsequently, the literature showed that pedestrians who have already used e-PMVs are less frightened during an encounter. The direction and speed are essential when e-kick scooters are approaching the pedestrian and invading their space. Other studies analyzed the driver's safety, evaluating the use of personal protective equipment (such as helmets) and experimenting with several devices to understand (safe/unsafe) driving behavior. These studies showed that social networks and advertisements do not emphasize helmets or protective devices. Furthermore, the literature indicates that new technologies are being developed to understand how users drive. Studies also showed that the main dynamics of crashes are related to the single driver losing his balance or colliding with the public space infrastructure. Finally, studies on the environmental impact were mainly conducted in Europe (perhaps also due to greater attention to environmental issues). E-PMVs have zero emissions while driving. However, to be considered sustainable and zero-impact, studies showed that their entire life cycle must be considered, and e-PMVs should be used for at least 9.5 months.

Although this survey analyzed separate issues, overlapping research questions emerged. For instance, the creation of an accessible national database with hospital and crashes detection data would be useful in both transport planning and crashes occurrence and severity prediction models; the analysis of travel models would be useful to understand which paths are most used and, thus, to provide input for the design of new infrastructures. Finally, as some issues of e-PMVs have been examined from a general viewpoint, more detailed reviews on separate issues are recommended.

Author Contributions: The authors confirm contribution to the paper as follows: study conception B.B., S.B., G.M.; methodology B.B.; S.B., formal analysis, S.B.; writing—original draft preparation, S.B.; writing—review and editing, B.B., G.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors would like to thank Michela Bonera, Martina Carra, Roberto Ventura, and David Vetturi for very helpful comments on earlier versions of this paper. The authors would like to thank the AE and anonymous reviewers for their very helpful suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A Overview on Publications between August and December 2020—Not Reviewed

Al Mamun, A., Zainol, N. R., & Hayat, N. (2020). Electric Scooter-An Alternative Mode of Transportation for Malaysian Youth.

Baek, K., Lee, H., Chung, J. H., & Kim, J. (2020). Electric scooter sharing: How do people value it as a last-mile transportation mode? Transportation Research Part D: Transport and Environment, 90, 102642.

Bell, J., Rogers, S., Mathew, J., Li, H., & Bullock, D. (2020, August). Comparing Speed Distribution of Micromobility Modes. In International Conference on Transportation and Development 2020 (pp. 59–67). Reston, VA: American Society of Civil Engineers.

Bieliński, T., & Ważna, A. (2020). Electric Scooter Sharing and Bike Sharing User Behavior and Characteristics. Sustainability, 12(22), 9640.

Button, K., Frye, H., & Reaves, D. (2020). Economic regulation and E-scooter networks in the USA. Research in Transportation Economics, 100973.

Cicchino, J. B., Kulie, P. E., & McCarthy, M. L. (2020). Severity of e-scooter rider injuries associated with trip characteristics.

Comer, A., Apathy, N., Waite, C., Bestmann, Z., Bradshaw, J., Burchfield, E., & Sabec, M. (2020). Electric Scooters (e-scooters): Assessing the Threat to Public Health and Safety in Setting Policies: Assessing e-scooter policies. Chronicles of Health Impact Assessment, 5 (1).

Curl, A., & Fitt, H. (2020). Same same, but different? Cycling and e-scootering in a rapidly changing urban transport landscape. New Zealand Geographer. (in press)

Dill, J., & McNeil, N. (2020). Are Shared Vehicles Shared by All? A Review of Equity and Vehicle Sharing. Journal of Planning Literature, 0885412220966732. (in press)

Douglass, K., Sikka, N., Boniface, K., Bhatt, K., McCarville, P., & Pourmand, A. (2020). Epidemiological Analysis of E-Scooter Injuries among Patients Presenting to the Emergency Department. Annals of Emergency Medicine, 76(4), S108.

English, K. C., Allen, J. R., Rix, K., Zane, D. F., Ziebell, C. M., Brown, C. V., & Brown, L. H. (2020). The characteristics of dockless electric rental scooter-related injuries in a large US city. Traffic injury prevention, 21(7), 476–481.

Faraji, F., Lee, J. H., Faraji, F., MacDonald, B., Oviedo, P., Stuart, E., & Castillo, E. M. (2020). Electric scooter craniofacial trauma. Laryngoscope Investigative Otolaryngology, 5(3), 390–395.

Farley, K. X., Aizpuru, M., Wilson, J. M., Daly, C. A., Xerogeanes, J., Gottschalk, M. B., & Wagner, E. R. (2020). Estimated incidence of electric scooter injuries in the US from 2014 to 2019. JAMA network open, 3(8), e2014500–e2014500.

Fearnley, N. (2020). Micromobility–Regulatory challenges and opportunities. In Shaping Smart Mobility Futures: Governance and Policy Instruments in times of Sustainability Transitions. Emerald Publishing Limited.

Feng, Y., Zhong, D., Sun, P., Zheng, W., Cao, Q., Luo, X., & Lu, Z. (2020). Micromobility in Smart Cities: A Closer Look at Shared Dockless E-Scooters via Big Social Data. arXiv preprint arXiv:2010.15203.

Gioldasis, C., & Christoforou, Z. (2020). Smart Infrastructure for Shared Mobility. In Conference on Sustainable Urban Mobility (pp. 970–979). Springer, Cham.

Glenn, J., Bluth, M., Christianson, M., Pressley, J., Taylor, A., Macfarlane, G. S., & Chaney, R. A. (2020). Considering the potential health impacts of electric scooters: an analysis of user reported behaviors in Provo, Utah. International journal of environmental research and public health, 17(17), 6344.

Glöss, M., Tuncer, S., Brown, B., Laurier, E., Pink, S., Fors, V., & Strömberg, H. (2020). New Mobilities: A Workshop on Mobility Beyond the Car. In Extended Abstracts of the 2020 CHI Conference on Human Factors in Computing Systems (pp. 1–8).

Hennocq, Q., Schouman, T., Khonsari, R. H., Sigaux, N., Descroix, V., Bertolus, C., & Foy, J. P. (2020). Evaluation of electric scooter head and neck injuries in Paris, 2017–2019. JAMA network open, 3(11), e2026698–e2026698.

Hosseinzadeh, A., Algomaiah, M., Kluger, R., & Li, Z. (2020). E-scooters and Sustainability: Investigating the Relationship between the Density of E-Scooter Trips and Characteristics of Sustainable Urban Development. Sustainable Cities and Society, 102624.

Johnston, K., Oakley, D., Durham, A. V., Bass, C., & Kershner, S. (2020). Regulating Micromobility: Examining Transportation Equity and Access. JCULP, 4, 682.

Kazmaier, M., Taefi, T. T., & Hettesheimer, T. (2020). Techno-Economical and Ecological Potential of Electric Scooters: A Life Cycle Analysis. European Journal of Transport and Infrastructure Research, 20(4), 233–251.

Kim, M., Lee, S., Ko, D. R., Kim, D. H., Huh, J. K., & Kim, J. Y. (2020). Craniofacial and dental injuries associated with stand-up electric scooters. Dental traumatology. (in press)

Köhler, S., Norkauer, A., Schmidt, M., & Loidl, V. (2020). Electrified Ultralight Vehicles as a Key Element for Door-to-Door Solutions in Urban Areas. In Innovations for Metropolitan Areas (pp. 65–76). Springer, Berlin, Heidelberg.

Kostareli, A., Basbas, S., Stamatiadis, N., & Nikiforiadis, A. (2020). Attitudes of E-Scooter Non-users Towards Users. In Conference on Sustainable Urban Mobility (pp. 87–96). Springer, Cham.

Laa, B., & Leth, U. (2020). Survey of E-scooter users in Vienna: Who they are and how they ride. Journal of Transport Geography, 89, 102874.

Lee, K. C., Naik, K., Wu, B. W., Karlis, V., Chuang, S. K., & Eisig, S. B. (2020). Are motorized scooters associated with more severe craniomaxillofacial injuries?. Journal of Oral and Maxillofacial Surgery.

Lipovsky, C. (2020). Free-floating electric scooters: representation in French mainstream media. International Journal of Sustainable Transportation, 1–10. (in press)

Liu, K. H., & Wang, S. M. (2020). Design and Evaluation of Electric Scooter Innovative Service. In 2020 IEEE International Conference on Consumer Electronics-Taiwan (ICCE-Taiwan) (pp. 1–2). IEEE.

Lo, D., Mintrom, C., Robinson, K., & Thomas, R. (2020). Shared micromobility: The influence of regulation on travel mode choice. New Zealand Geographer.

Löcken, A., Brunner, P., & Kates, R. (2020). Impact of Hand Signals on Safety: Two Controlled Studies With Novice E-Scooter Riders. In 12th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (pp. 132–140).

Lorne Platt & Greg Rybarczyk (2020) Skateboarder and scooter-rider perceptions of the urban environment: a qualitative analysis of user-generated content, Urban Geography, DOI: 10.1080/02723638.2020.1811554

Martínez-Navarro, A., Cloquell-Ballester, V. A., & Segui-Chilet, S. (2020). Photovoltaic Electric Scooter Charger Dock for the Development of Sustainable Mobility in Urban Environments. IEEE Access, 8, 169486–169495. (in press)

Moftakhar, T., Wanzel, M., Vojcsik, A., Kralinger, F., Mousavi, M., Hajdu, S., & Starlinger, J. (2020). Incidence and severity of electric scooter related injuries after introduction of an urban rental programme in Vienna: a retrospective multicentre study. Archives of orthopaedic and trauma surgery, 1–7. (in press)

Oeschger, G., Carroll, P., & Caulfield, B. (2020). Micromobility and public transport integration: The current state of knowledge. Transportation Research Part D: Transport and Environment, 89, 102628.

Oksanen, E., Turunen, A., & Thorén, H. (2020). Assessment of Craniomaxillofacial Injuries Following Electric Scooter Crashes in Turku, Finland in 2019. Journal of Oral and Maxillofacial Surgery.

Rahman, M., Garcia-Ballestas, E., Agrawal, A., Moscote-Salazar, L.R. (2020). E-scooter and neurotrauma. Neurochirurgie, doi:10.1016/j.neuchi.2020.09.004. (in press)

Raptopoulou, A., Basbas, S., Stamatiadis, N., & Nikiforiadis, A. (2020). A First Look at E-Scooter Users. In Conference on Sustainable Urban Mobility (pp. 882–891). Springer, Cham.

Reck, D. J., Guidon, S., Haitao, H., & Axhausen, K. W. (2020). Explaining shared micromobility usage, competition and mode choice by modelling empirical data from Zurich, Switzerland. Transportation Research Part C: Emerging Technologies.

Riggs, W., & Kawashima, M. (2020). Exploring Best Practice for Municipal E-Scooter Policy in the United States. Available at SSRN 3512725.

Rix, K., Demchur, N. J., Zane, D. F., & Brown, L. H. (2020). Injury rates per mile of travel for electric scooters versus motor vehicles. The American Journal of Emergency Medicine. (in press)

Ruhrort, L. (2020). Reassessing the role of shared mobility services in a transport transition: Can they contribute the rise of an alternative socio-technical regime of mobility?. Sustainability, 12(19), 8253.

Saum, N., Sugiura, S., & Piantanakulchai, M. (2020). Short-Term Demand and Volatility Prediction of Shared Micromobility: a case study of e-scooter in Thammasat University. 2020 Forum on Integrated and Sustainable Transportation Systems (FISTS), Delft, South Holland Province, Netherlands, pp. 27–32, doi: 10.1109/FISTS46898.2020.9264852. Shiffler, K., Mancini, K., Wilson, M., Huang, A., Mejia, E., & Yip, F. K. (2020). Intoxication is a significant risk factor for severe craniomaxillofacial injuries in standing electric scooter crashes. Journal of Oral and Maxillofacial Surgery. (in press)

Siow, M. Y., Lavoie-Gagne, O., Politzer, C. S., Mitchell, B. C., Harkin, W. E., Flores, A. R., & Kent, W. T. (2020). Electric scooter orthopaedic injury demographics at an urban Level I trauma center. Journal of orthopaedic trauma, 34(11), e424–e429.

Smit, R. B., Graham, D. O., & Erasmus, J. (2020). E-scooter injuries referred to the Oral and Maxillofacial Surgical Service at Christchurch Hospital: A retrospective observational study and cost-analysis of 17-months of data. British Journal of Oral and Maxillofacial Surgery.

Sokolowski, M. M. (2020). Laws and Policies on Electric Scooters in the European Union: A Ride to the Micromobility Directive?. European Energy and Environmental Law Review, 29(4).

SooHoo, C., & SooHoo, J. (2020). Importance of educating teenagers on appropriate safety gear for e-scooters. comment on" follow-up investigation on the promotional practices of electric scooter companies: content analysis of posts on Instagram and Twitter". JMIR public health and surveillance, 6(4), e18945–e18945.

Štraub, D., & Gajda, A. (2020). E-scooter sharing schemes operational zones in Poland: dataset on voivodeship capital cities. Data in Brief, 106560.

Sundqvist-Andberg, H., Auvinen, H., & Tuominen, A. (2020). Sustainability of Electric Scooter Sharing Services. In XXXI ISPIM Innovation Conference: Innovating in Times of Crises.

Tan, S., & Tamminga, K. (2020). A Vision for Urban Micromobility: From Current Streetscape to City of the Future. In Conference on Sustainable Urban Mobility (pp. 158–167). Springer, Cham.

Vorina, A., Čakš, G., & Županec, J. (2020). GREEN TRANSPORT–THE USAGE OF ELECTRIC SCOOTER FOR PUBLIC TRANSPORT IN TOWNS.

Watson, H. N., Garman, C. M., Wishart, J., & Zimmermann, J. (2020). Patient Demographics and Injury Characteristics of ER Visits Related to Powered-Scooters (No. 2020-01-0933). SAE Technical Paper.

Wüster, J., Voß, J., Koerdt, S., Beck-Broichsitter, B., Kreutzer, K., Märdian, S., & Doll, C. (2020). Impact of the Rising Number of Rentable E-scooter Crashes on Emergency Care in Berlin 6 Months after the Introduction: A Maxillofacial Perspective. Craniomaxillofacial Trauma & Reconstruction, 1943387520940180.

References

- 1. POLIS. Macro Managing Micro Mobility. Taking the Long View on Short Trips; POLIS: Brussels, Belgium, 2019.
- 2. Ambrosino, G.; Liberato, A.; Pettinelli, I. Sustainable Urban Logistics Plans (SULP) Guidelines. Available online: https://civitas. eu/sites/default/files/documents/sulp_guidlines.pdf (accessed on 4 January 2021).
- 3. SUMP. *Guidelines. Developing and Implementing a Sustainable Urban Mobility Plan,* 2nd ed.; European Commission: Brussels, Belgium, 2019.
- 4. Taniguchi, E. City Logistics for Sustainable and Liveable Cities. In *Green Logistics and Transportation*; Fahimnia, B., Bell, M.G.H., Hensher, D.A., Sarkis, J., Eds.; Springer International Publishing: Cham, Switzerland, 2015; pp. 49–60. ISBN 978-3-319-17180-7.
- Russo, F.; Comi, A. Investigating the Effects of City Logistics Measures on the Economy of the City. Sustainability 2020, 12, 1439. [CrossRef]
- 6. Luo, J.; Boriboonsomins, K.; Barth, M. Consideration of Exposure to Traffic-Related Air Pollution in Bicycle Route Planning. J. *Transp. Health* **2020**, *16*, 100792. [CrossRef]
- 7. The International Transport Forum. Safe Micromobility; OECD: Paris, France, 2020; p. 98.
- ReportBuyer. Personal Mobility Devices Market-Global Industry Analysis, Size, Share, Growth, Trends, and Forecast, 2019–2027. Available online: https://www.reportbuyer.com/product/4142781/personal-mobility-devices-market-global-industry-analysissize-share-growth-trends-and-forecast-2019-2027.html?utm_source=PRN (accessed on 7 April 2020).
- 9. Chang, A.Y.; Miranda-Moreno, L.; Clewlow, R.; Sun, L. *Trend or Fad? Deciphering the Enablers of Micromobility in the U.S.*; SAE International: Warrendale, PA, USA, 2019.
- 10. Lawrence, M. Experiential Graphic Design: Generating Urban Renewal by Improving Safety and Connectivity. Master's Thesis, Kent State University, Kent, OH, USA, 2016.
- 11. O'Hern, S.; Estgfaeller, N. A Scientometric Review of Powered Micromobility. Sustainability 2020, 12, 9505. [CrossRef]

- 12. Cooper, H.M. Integrating Research: A Guide for Literature Reviews; SAGE Publications: Newbury Park, CA, USA, 1989.
- 13. Moher, D.; Liberati, A.; Tetzlaff, J.; Altman, D.G. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. *PLoS Med.* **2019**, *6*, e1000097.
- 14. Barabino, B.; Lai, C.; Olivo, A. Fare Evasion in Public Transport Systems: A Review of the Literature. *Public Transp.* **2020**, *12*, 27–88. [CrossRef]
- 15. Kurrer, C. Politica Ambientale: Principi Generali e Quadro Di Riferimento. Available online: https://www.europarl.europa.eu/ factsheets/it/sheet/71/politica-ambientale-principi-generali-e-quadro-di-riferimento (accessed on 4 January 2021).
- 16. Almannaa, M.H.; Ashqar, H.I.; Elhenawy, M.; Masoud, M.; Rakotonirainy, A.; Rakha, H. A Comparative Analysis of E-Scooter and e-Bike Usage Patterns: Findings from the City of Austin, TX. *Int. J. Sustain. Transp.* **2020**, 1–9. [CrossRef]
- 17. Bai, S.; Jiao, J. Dockless E-Scooter Usage Patterns and Urban Built Environments: A Comparison Study of Austin, TX, and Minneapolis, MN. *Travel Behav. Soc.* 2020, 20, 264–272. [CrossRef]
- Caspi, O.; Smart, M.J.; Noland, R.B. Spatial Associations of Dockless Shared E-Scooter Usage. Transp. Res. Part D Transp. Environ. 2020, 86, 102396. [CrossRef] [PubMed]
- 19. Davies, N.; Blazejewski, L.; Sherriff, G. The Rise of Micromobilities at Tourism Destinations. JTF 2020, 6, 209–212. [CrossRef]
- 20. Espinoza, W.; Howard, M.; Lane, J.; Van Hentenryck, P. Shared E-Scooters: Business, Pleasure, or Transit? *arXiv* 2019, arXiv:1910.05807.
- Hardt, C.; Bogenberger, K. Usability of Escooters in Urban Environments—A Pilot Study. In Proceedings of the 2017 IEEE Intelligent Vehicles Symposium (IV), Los Angeles, CA, USA, 11–14 June 2017; IEEE: Los Angeles, CA, USA, June, 2017; pp. 1650–1657.
- 22. Hardt, C.; Bogenberger, K. Usage of E-Scooters in Urban Environments. Transp. Res. Procedia 2019, 37, 155–162. [CrossRef]
- Hawa, L.; Cui, B.; Sun, L.; El-Geneidy, A. Scoot Over: Determinants of Shared Electric Scooter Use in Washington D.C. In Proceedings of the Transportation Research Board 99th Annual Meeting, Washington, DC, USA, 12–16 January 2020.
- 24. Jiao, J.; Bai, S. Understanding the Shared E-Scooter Travels in Austin, TX. IJGI 2020, 9, 135. [CrossRef]
- 25. Krizek, K.J.; McGuckin, N. Shedding NHTS Light on the Use of 'Little Vehicles' in Urban Areas. *Transp. Find.* 2019. [CrossRef]
- Markvica, K.; Schwieger, K.; Aleksa, M. E-Scooter as Environmentally Friendly Last Mile Option? Insights on Spatial and Infrastructural Implications for Urban Areas Based on the Example of Vienna. In Proceedings of the Real Corp 2020: Shaping Urban Change, Vienna, Austria, 15–18 September 2020.
- 27. Mathew, J.K.; Liu, M.; Seeder, S.; Li, H.; Bullock, D.M. Analysis of E-Scooter Trips and Their Temporal Usage Patterns. *ITE J. Inst. Trasp. Eng.* **2019**, *89*, 44–49.
- Mathew, J.K.; Liu, M.; Bullock, D.M. Impact of Weather on Shared Electric Scooter Utilization. In Proceedings of the 2019 IEEE Intelligent Transportation Systems Conference (ITSC), Auckland, New Zealand, 27–30 October 2019; IEEE: Auckland, New Zealand, October 2019; pp. 4512–4516.
- McKenzie, G. Shared Micro-Mobility Patterns as Measures of City Similarity: Position Paper. In Proceedings of the 1st ACM SIGSPATIAL International Workshop on Computing with Multifaceted Movement Data—MOVE'19; ACM Press: Chicago, IL, USA, 2019; pp. 1–4.
- 30. McKenzie, G. Spatiotemporal Comparative Analysis of Scooter-Share and Bike-Share Usage Patterns in Washington, D.C. J. *Transp. Geogr.* **2019**, *78*, 19–28. [CrossRef]
- 31. McKenzie, G. Urban Mobility in the Sharing Economy: A Spatiotemporal Comparison of Shared Mobility Services. *Comput. Environ. Urban Syst.* 2020, 79, 101418. [CrossRef]
- 32. Nocerino, R.; Colorni, A.; Lia, F.; Luè, A. E-Bikes and E-Scooters for Smart Logistics: Environmental and Economic Sustainability in Pro-E-Bike Italian Pilots. *Transp. Res. Procedia* 2016, 14, 2362–2371. [CrossRef]
- 33. Noland, R.B. Trip Patterns and Revenue of Shared E-Scooters in Louisville, Kentucky. Transp. Find. 2019. [CrossRef]
- 34. Reck, D.J.; Guidon, S.; Haitao, H.; Axhausen, K.W. Shared Micromobility in Zurich, Switzerland: Analysing Usage, Competition and Mode Choice. In 20th Swiss Transport Research Conference (STRC 2020); IVT, ETH Zurich: Zürich, Switzerland, 2020. [CrossRef]
- 35. Tuncer, S.; Brown, B. E-Scooters on the Ground: Lessons for Redesigning Urban Micro-Mobility. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems; ACM: Honolulu, HI, USA, 21 April 2020; pp. 1–14.
- Zou, Z.; Younes, H.; Erdoğan, S.; Wu, J. Exploratory Analysis of Real-Time E-Scooter Trip Data in Washington, D.C. Transp. Res. Rec. 2020, 2674, 285–299. [CrossRef]
- 37. Berge, S.H. Kickstarting Micromobility—A Pilot Study on e-Scooters; Institute of Transport Economics: Oslo, Norway, 2019; Volume 4.
- Hyvönen, K.; Repo, P.; Lammi, M. Light Electric Vehicles: Substitution and Future Uses. *Transp. Res. Procedia* 2016, 19, 258–268.
 [CrossRef]
- 39. Lee, M.; Chow, J.Y.J.; Yoon, G.; He, B.Y. Forecasting E-Scooter Competition with Direct and Access Trips by Mode and Distance in New York City. *arXiv* **2019**, arXiv:1908.08127.
- Tomita, K.; Hashimoto, N.; Kamimura, A.; Yokozuka, M.; Matsumoto, O. Experimental Examination and Simulation Analysis of Standing-Type Personal Mobility Device Sharing. In Proceedings of the 2016 19th Conference of Open Innovations Association (FRUCT), Jyvaskyla, Finland, 7–11 November 2016; IEEE: Jyvaskyla, Finland, 2016; pp. 248–255.
- 41. EIT Urban Mobility. *Guideline of Best Practices, and Results of e-Micromobile Integration Potentials;* EIT Urban Mobility: Stockholm, Sweden, 2020; p. 112.

- Degele, J.; Gorr, A.; Haas, K.; Kormann, D.; Krauss, S.; Lipinski, P.; Tenbih, M.; Koppenhoefer, C.; Fauser, J.; Hertweck, D. Identifying E-Scooter Sharing Customer Segments Using Clustering. In Proceedings of the 2018 IEEE International Conference on Engineering, Technology and Innovation (ICE/ITMC), Stuttgard, Germany, 17–20 June 2018; pp. 1–8.
- 43. Eccarius, T.; Lu, C.-C. Adoption Intentions for Micro-Mobility—Insights from Electric Scooter Sharing in Taiwan. *Transp. Res. Part* D *Transp. Environ.* **2020**, *84*, 102327. [CrossRef]
- 44. Fitt, H.; Curl, A. The Early Days of Shared Micromobility: A Social Practices Approach. J. Transp. Geogr. 2020, 86, 102779. [CrossRef]
- 45. Huang, F.-H.; Lin, S.-R. A Survey of User Experience of Two Wheeler Users in Long-Term Interactions. In Proceedings of the 20th Congress of the International Ergonomics Association (IEA 2018); Bagnara, S., Tartaglia, R., Albolino, S., Alexander, T., Fujita, Y., Eds.; Advances in Intelligent Systems and Computing. Springer International Publishing: Cham, Switzerland, 2019; Volume 824, pp. 1465–1472, ISBN 978-3-319-96070-8.
- 46. Sanders, R.L.; Branion-Calles, M.; Nelson, T.A. To Scoot or Not to Scoot: Findings from a Recent Survey about the Benefits and Barriers of Using E-Scooters for Riders and Non-Riders. *Transp. Res. Part A Policy Pract.* **2020**, 139, 217–227. [CrossRef]
- Arellano, J.; Fang, K. Sunday Drivers, or Too Fast and Too Furious? Analyzing Speed and Rider Behaviour of E-Scooter Riders in San Jose, California (2nd Highest Scoring Masters/Undergraduate Abstract Award Sponsored by HNTB Corporation—Great Lakes Region). J. Transp. Health 2019, 14, 100725. [CrossRef]
- 48. Brown, A.; Klein, N.J.; Thigpen, C.; Williams, N. Impeding Access: The Frequency and Characteristics of Improper Scooter, Bike, and Car Parking. *Transp. Res. Interdiscip. Perspect.* **2020**, *4*, 100099. [CrossRef]
- 49. Fang, K.; Agrawal, A.W.; Steele, J.; Hunter, J.J.; Hooper, A.M. Where Do Riders Park Dockless, Shared Electric Scooters? Findings from San Jose, California; Mineta Transportation Institute Publications: San José, CA, USA, 2018; Volume 6.
- 50. James, O.; Swiderski, J.; Hicks, J.; Teoman, D.; Buehler, R. Pedestrians and E-Scooters: An Initial Look at E-Scooter Parking and Perceptions by Riders and Non-Riders. *Sustainability* **2019**, *11*, 5591. [CrossRef]
- 51. Nishiuchi, H.; Shiomi, Y.; Todoroki, T. Segway Running Behavior Focusing on Riders' Experience Based on Image-Processing Data. *Asian Transp. Stud.* **2015**, *3*, 467–468.
- 52. Tuncer, S.; Laurier, E.; Brown, B.; Licoppe, C. Notes on the Practices and Appearances of E-Scooter Users in Public Space. *J. Transp. Geogr.* 2020, *85*, 102702. [CrossRef]
- 53. Butrina, P.; Le Vine, S.; Henao, A.; Sperling, J.; Young, S.E. Municipal Adaptation to Changing Curbside Demands: Exploratory Findings from Semi-Structured Interviews with Ten U.S. Cities. *Transp. Policy* **2020**, *92*, 1–7. [CrossRef]
- 54. Chen, Y.-W.; Cheng, C.-Y.; Li, S.-F.; Yu, C.-H. Location Optimization for Multiple Types of Charging Stations for Electric Scooters. *Appl. Soft Comput.* **2018**, *67*, 519–528. [CrossRef]
- 55. Clark, A.V.; Atkinson-Palombo, C.; Garrick, N.W. The Rise and Fall of the Segway. Transfers 2019, 9, 27–44. [CrossRef]
- Fang, S.-C.; Chang, I.-C.; Yu, T.-Y. Assessment of the Behavior and Characteristics of Electric Scooter Use on Islands. J. Clean. Prod. 2014, 108, 1193–1202. [CrossRef]
- 57. Gössling, S. Integrating E-Scooters in Urban Transportation: Problems, Policies, and the Prospect of System Change. *Transp. Res. Part D Transp. Environ.* **2020**, *79*, 102230. [CrossRef]
- Janssen, C.; Barbour, W.; Hafkenschiel, E.; Abkowitz, M.; Philip, C.; Work, D.B. City-to-City and Temporal Assessment of Peer City Scooter Policy. *Transp. Res. Rec.* 2020, 2674, 219–232. [CrossRef]
- Jiménez, D.; De La Fuente, Y.; Hernández-Galán, J. Diversity of "Pedestrians on Wheels", New Challenges for Cities in 21st Century. Stud. Health Technol. Inform. 2018, 10. [CrossRef]
- 60. Lo, D.; Mintrom, C.; Robinson, K.; Thomas, R. Shared Micromobility: The Influence of Regulation on Travel Mode Choice. N. Z. *Geogr.* 2020, *76*, 135–146. [CrossRef]
- 61. Moran, M.E.; Laa, B.; Emberger, G. Six Scooter Operators, Six Maps: Spatial Coverage and Regulation of Micromobility in Vienna, Austria. *Case Stud. Transp. Policy* 2020, *8*, 658–671. [CrossRef]
- 62. Badeau, A.; Carman, C.; Newman, M.; Steenblik, J.; Carlson, M.; Madsen, T. Emergency Department Visits for Electric Scooter-Related Injuries after Introduction of an Urban Rental Program. *Am. J. Emerg. Med.* **2019**, *37*, 1531–1533. [CrossRef]
- 63. Beck, S.; Barker, L.; Chan, A.; Stanbridge, S. Emergency Department Impact Following the Introduction of an Electric Scooter Sharing Service. *Emerg. Med. Australas.* **2020**, *32*, 409–415. [CrossRef]
- 64. Bekhit, M.N.Z.; Le Fevre, J.; Bergin, C.J. Regional Healthcare Costs and Burden of Injury Associated with Electric Scooters. *Injury* 2020, *51*, 271–277. [CrossRef]
- 65. Blomberg, S.N.F.; Rosenkrantz, O.C.M.; Lippert, F.; Collatz Christensen, H. Injury from Electric Scooters in Copenhagen: A Retrospective Cohort Study. *BMJ Open* **2019**, *9*, e033988. [CrossRef]
- 66. Bloom, M.B.; Noorzad, A.; Lin, C.; Little, M.; Lee, E.Y.; Margulies, D.R.; Torbati, S.S. Standing Electric Scooter Injuries: Impact on a Community. *Am. J. Surg.* 2020, 221, 227–232. [CrossRef]
- 67. Dhillon, N.K.; Juillard, C.; Barmparas, G.; Lin, T.-L.; Kim, D.Y.; Turay, D.; Seibold, A.R.; Kaminski, S.; Duncan, T.K.; Diaz, G.; et al. Electric Scooter Injury in Southern California Trauma Centers. *J. Am. Coll. Surg.* **2020**, 231, 133–138. [CrossRef]
- 68. Haworth, N.L.; Schramm, A. Illegal and Risky Riding of Electric Scooters in Brisbane. *Med. J. Aust.* 2019, 211, 412–413. [CrossRef] [PubMed]

- 69. Ishmael, C.R.; Hsiue, P.P.; Zoller, S.D.; Wang, P.; Hori, K.R.; Gatto, J.D.; Li, R.; Jeffcoat, D.M.; Johnson, E.E.; Bernthal, N.M. An Early Look at Operative Orthopaedic Injuries Associated with Electric Scooter Accidents: Bringing High-Energy Trauma to a Wider Audience. *J. Bone Jt. Surg.* 2020, *102*, e18. [CrossRef] [PubMed]
- Islam, A.; Koger, K.; VandenBerg, S.; Wang, D.; Lang, E. The Summer of the E-Scooter: A Multicenter Evaluation of the Emergency Department Impact of Rentable Motorized Scooters in Calgary. CJEM 2020, 22, S93. [CrossRef]
- Kim, Y.W.; Park, W.B.; Cho, J.S.; Hyun, S.Y.; Lee, G. The New Recreational Transportation on the Street: Personal Mobility, Is It Safe? J. Trauma Inj. 2018, 31, 125–134. [CrossRef]
- 72. King, C.S.; Liu, M.; Patel, S.; Goo, T.; Lim, W.; Toh, H. Injury Patterns Associated with Personal Mobility Devices and Electric Bicycles: An Analysis from an Acute General Hospital in Singapore. *Singap. Med. J.* **2020**, *61*, 96–101. [CrossRef] [PubMed]
- 73. Kobayashi, L.M.; Williams, E.; Brown, C.V.; Emigh, B.J.; Bansal, V.; Badiee, J.; Checchi, K.D.; Castillo, E.M.; Doucet, J. The E-Merging e-Pidemic of e-Scooters. *Trauma Surg. Acute Care Open* **2019**, *4*, e000337. [CrossRef] [PubMed]
- 74. Liew, Y.; Wee, C.; Pek, J. New Peril on Our Roads: A Retrospective Study of Electric Scooter-Related Injuries. *Singap. Med. J.* 2020, 61, 92–95. [CrossRef] [PubMed]
- 75. Mayhew, L.J.; Bergin, C. Impact of E-scooter Injuries on Emergency Department Imaging. J. Med. Imaging Radiat. Oncol. 2019, 63, 461–466. [CrossRef] [PubMed]
- 76. Mitchell, G.; Tsao, H.; Randell, T.; Marks, J.; Mackay, P. Impact of Electric Scooters to a Tertiary Emergency Department: 8-week Review after Implementation of a Scooter Share Scheme. *Emerg. Med. Australas.* **2019**, *31*, 930–934. [CrossRef] [PubMed]
- 77. Namiri, N.K.; Lui, H.; Tangney, T.; Allen, I.E.; Cohen, A.J.; Breyer, B.N. Electric Scooter Injuries and Hospital Admissions in the United States, 2014–2018. *JAMA Surg.* 2020, *155*, 357. [CrossRef] [PubMed]
- 78. Nellamattathil, M.; Amber, I. An Evaluation of Scooter Injury and Injury Patterns Following Widespread Adoption of E-Scooters in a Major Metropolitan Area. *Clin. Imaging* **2020**, *60*, 200–203. [CrossRef]
- 79. Pourmand, A.; Liao, J.; Pines, J.M.; Mazer-Amirshahi, M. Segway®Personal Transporter-Related Injuries: A Systematic Literature Review and Implications for Acute and Emergency Care. *J. Emerg. Med.* **2018**, *54*, 630–635. [CrossRef] [PubMed]
- 80. Puzio, T.J.; Murphy, P.B.; Gazzetta, J.; Dineen, H.A.; Savage, S.A.; Streib, E.W.; Zarzaur, B.L. The Electric Scooter: A Surging New Mode of Transportation That Comes with Risk to Riders. *Traffic Inj. Prev.* **2020**, *21*, 175–178. [CrossRef]
- 81. Roider, D.; Busch, C.; Spitaler, R.; Hertz, H. Segway®Related Injuries in Vienna: Report from the Lorenz Böhler Trauma Centre. *Eur. J. Trauma Emerg. Surg.* 2016, 42, 203–205. [CrossRef]
- Störmann, P.; Klug, A.; Nau, C.; Verboket, R.D.; Leiblein, M.; Müller, D.; Schweigkofler, U.; Hoffmann, R.; Marzi, I.; Lustenberger, T. Characteristics and Injury Patterns in Electric-Scooter Related Accidents—A Prospective Two-Center Report from Germany. *JCM* 2020, 9, 1569. [CrossRef]
- 83. Trivedi, T.K.; Liu, C.; Antonio, A.L.M.; Wheaton, N.; Kreger, V.; Yap, A.; Schriger, D.; Elmore, J.G. Injuries Associated With Standing Electric Scooter Use. *JAMA Netw. Open* **2019**, *2*, e187381. [CrossRef]
- 84. Vernon, N.; Maddu, K.; Hanna, T.N.; Chahine, A.; Leonard, C.E.; Johnson, J.-O. Emergency Department Visits Resulting from Electric Scooter Use in a Major Southeast Metropolitan Area. *Emerg. Radiol.* **2020**, *27*, 469–475. [CrossRef] [PubMed]
- 85. Che, M.; Lum, K.M.; Wong, Y.D. Users' Attitudes on Electric Scooter Riding Speed on Shared Footpath: A Virtual Reality Study. *Int. J. Sustain. Transp.* **2020**, *15*, 152–161. [CrossRef]
- 86. Hasegawa, Y.; Dias, C.; Iryo-Asano, M.; Nishiuchi, H. Modeling Pedestrians' Subjective Danger Perception toward Personal Mobility Vehicles. *Transp. Res. Part F Traffic Psychol. Behav.* **2018**, *56*, 256–267. [CrossRef]
- Kuo, J.-Y.; Sayeed, A.; Tangirala, N.T.; Chua Yi Han, V.; Dauwels, J.; Mayer, M.P. Pedestrians' Acceptance of Personal Mobility Devices on the Shared Path: A Structural Equation Modelling Approach. In Proceedings of the 2019 IEEE Intelligent Transportation Systems Conference (ITSC), Auckland, New Zealand, 27–30 October 2019; IEEE: Auckland, New Zealand, October 2019; pp. 2349–2354.
- Kuo, J.-Y.; Tangirala, N.T.; Murugesan, J.; Sayeed, A.; Chua, Y.H.V.; Dauwels, J.; Mayer, M.P. Experimental Analysis of Pedestrians' Discomfort Zone for Personal Mobility Devices on the Footpath. In Proceedings of the 2019 IEEE 90th Vehicular Technology Conference (VTC2019-Fall), Honolulu, HI, USA, 22–25 September 2019; IEEE: Honolulu, HI, USA, September 2019; pp. 1–5.
- 89. Maiti, A.; Vinayaga-Sureshkanth, N.; Jadliwala, M.; Wijewickrama, R.; Griffin, G.P. Impact of E-Scooters on Pedestrian Safety: A Field Study Using Pedestrian Crowd-Sensing. *arXiv* **2020**, arXiv:1908.05846.
- 90. Pham, T.Q. Investigation of Avoidance Assistance System for the Driver of a Personal Transporter Using Personal Space: A Simulation Based Study. *Int. J. Mech. Eng. Robot. Res.* **2019**, *8*, 254–259. [CrossRef]
- 91. Pham, T.Q.; Nakagawa, C.; Shintani, A.; Ito, T. Evaluation of the Effects of a Personal Mobility Vehicle on Multiple Pedestrians Using Personal Space. *IEEE Trans. Intell. Transport. Syst.* **2015**, *16*, 2028–2037. [CrossRef]
- 92. Hall, E.T. The Hidden Dimension, 2nd ed.; Doubleday and Company: New York, NY, USA, 1982; pp. 113–129.
- Allem, J.-P.; Majmundar, A. Are Electric Scooters Promoted on Social Media with Safety in Mind? A Case Study on Bird's Instagram. *Prev. Med. Rep.* 2019, 13, 62–63. [CrossRef] [PubMed]
- 94. Dormanesh, A.; Majmundar, A.; Allem, J.-P. Follow-Up Investigation on the Promotional Practices of Electric Scooter Companies: Content Analysis of Posts on Instagram and Twitter. *JMIR Public Health Surveill* **2020**, *6*, e16833. [CrossRef]
- Guo, Y.; Sayed, T.; Zaki, M.H. Examining Two-Wheelers' Overtaking Behavior and Lateral Distance Choices at a Shared Roadway Facility. J. Transp. Saf. Secur. 2020, 12, 1046–1066. [CrossRef]

- 96. Kim, J.; Sato, K.; Hashimoto, N.; Kashevnik, A.; Tomita, K.; Miyakoshi, S.; Takinami, Y.; Matsumoto, O.; Boyali, A. Impact of the Face Angle to Traveling Trajectory during the Riding Standing-Type Personal Mobility Device. In *MATEC Web of Conferences*; EDP Sciences: Les Ulis, France, 2018; Volume 161, p. 03001. [CrossRef]
- 97. Nisson, P.L.; Ley, E.; Chu, R. Electric Scooters: Case Reports Indicate a Growing Public Health Concern. *Am. J. Public Health* 2020, 110, 177–179. [CrossRef]
- Terrell, G.C. Characterizing Electric Scooter Riding Behavior to Detect Unsafe and Non-Compliant Use. *Tech. Discl. Commons* 2019. Available online: https://www.tdcommons.org/dpubs_series/2280 (accessed on 15 March 2020).
- 99. Kay, J.J.; Savolainen, P.T.; Gates, T.J.; Datta, T.K. Driver Behavior during Bicycle Passing Maneuvers in Response to a Share the Road Sign Treatment. *Accid. Anal. Prev.* **2014**, *70*, 92–99. [CrossRef]
- 100. Xu, J.; Shang, S.; Yu, G.; Qi, H.; Wang, Y.; Xu, S. Are Electric Self-Balancing Scooters Safe in Vehicle Crash Accidents? Accid. Anal. Prev. 2016, 87, 102–116. [CrossRef] [PubMed]
- Xu, J.; Shang, S.; Qi, H.; Yu, G.; Wang, Y.; Chen, P. Simulative Investigation on Head Injuries of Electric Self-Balancing Scooter Riders Subject to Ground Impact. Accid. Anal. Prev. 2016, 89, 128–141. [CrossRef] [PubMed]
- 102. Yang, H.; Ma, Q.; Wang, Z.; Cai, Q.; Xie, K.; Yang, D. Safety of Micro-Mobility: Analysis of E-Scooter Crashes by Mining News Reports. Accid. Anal. Prev. 2020, 143, 105608. [CrossRef] [PubMed]
- 103. de Bortoli, A.; Christoforou, Z. Consequential LCA for Territorial and Multimodal Transportation Policies: Method and Application to the Free-Floating e-Scooter Disruption in Paris. J. Clean. Prod. 2020, 273, 122898. [CrossRef]
- Hollingsworth, J.; Copeland, B.; Johnson, J.X. Are E-Scooters Polluters? The Environmental Impacts of Shared Dockless Electric Scooters. Environ. Res. Lett. 2019, 14, 084031. [CrossRef]
- Moreau, H.; de Jamblinne de Meux, L.; Zeller, V.; D'Ans, P.; Ruwet, C.; Achten, W.M.J. Dockless E-Scooter: A Green Solution for Mobility? Comparative Case Study between Dockless E-Scooters, Displaced Transport, and Personal E-Scooters. *Sustainability* 2020, 12, 1803. [CrossRef]
- 106. Severengiz, S.; Finke, S.; Schelte, N.; Wendt, N. Life Cycle Assessment on the Mobility Service E-Scooter Sharing. In Proceedings of the 2020 IEEE European Technology and Engineering Management Summit (E-TEMS), Dortmund, Germany, 5–7 March 2020; IEEE: Dortmund, Germany, March 2020; pp. 1–6.
- 107. Porcu, F.; Olivo, A.; Maternini, G.; Barabino, B. Evaluating Bus Accident Risks in Public Transport. *Transp. Res. Procedia* 2020, 45, 443–450. [CrossRef]